

Temi di Discussione

(Working Papers)

The macroeconomics of fiscal consolidations in a monetary union: the case of Italy

by Lorenzo Forni, Andrea Gerali and Massimiliano Pisani







Temi di discussione

(Working papers)

The macroeconomics of fiscal consolidations in a monetary union: the case of Italy

by Lorenzo Forni, Andrea Gerali and Massimiliano Pisani

Number 747 - March 2010

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Alfonso Rosolia, Marcello Pericoli, Ugo Albertazzi, Daniela Marconi, Andrea Neri, Giulio Nicoletti, Paolo Pinotti, Marzia Romanelli, Enrico Sette, Fabrizio Venditti. *Editorial Assistants:* Roberto Marano, Nicoletta Olivanti.

THE MACROECONOMICS OF FISCAL CONSOLIDATIONS IN A MONETARY UNION: THE CASE OF ITALY

by Lorenzo Forni*, Andrea Gerali* and Massimiliano Pisani*

Abstract

We simulate the macroeconomic and welfare implications of different fiscal consolidation scenarios in Italy using a medium scale two-areas dynamic general equilibrium currencyunion model. Differently from similar models, ours is rich in the terms of fiscal features. We assume distortionary taxes (on labor income, capital income and consumption) and welfareenhancing public expenditure. We distinguish between public spending on final goods and services, public employment and transfers to households. The scenarios that we consider envisage a decreases in the public debt to GDP ratio of 10 percentage points in 5 years. Based on our simulations we find that: first, fiscal distortions are quantitatively significant; second, a consolidation strategy that reduces expenditure and simultaneously lowers tax rates has a positive effect on long-run GDP of 5% to 7% and on welfare of 4% to 7% of the initial levels, depending on the composition of the adjustment; third, consumption and investment are stable or grow on impact and along the path to the new steady state; finally, spillovers to the rest of the Euro area are expansionary and sizeable both in the long run and along the transition.

JEL Classification: E62, H63.

Keywords: fiscal consolidation, monetary union, distortionary taxation, general equilibrium models.

	C CHITCHICS	
1. Introducti	O n	5
2. The mode	1	
2.1 The s	etup	
2.2 Fisca	policy	9
2.3 House	eholds	11
2.4 Calib	ration	12
3. Results		14
3.1 Optin	nal expenditure and revenue composition for given level of public debt	15
3.2 The l	ong-run effects of the fiscal consolidation	18
4. Transition	dynamics	20
5. Robustnes	S	21
6. Concludir	g remarks	23
References		24
Appendix		26
Tables and fi	gures	34

Contents

^{*} Bank of Italy, Economics, Research and International Relations Division.

1 Introduction¹

Recent forecasts by the European Commission and the International Monetary Fund point to dramatic increases in the level of public debt in the next few years in almost all euro area countries. The grim perspectives of the public accounts are compounded by the very high level of implicit public debt, related to the promises of the health and pension systems in our aging societies. Therefore, inevitably, in the near future will see a renewed debate on how to consolidate the fiscal position.

This paper contributes to the debate by simulating the macroeconomic and welfare implications of different fiscal consolidation scenarios in Italy. The model that we use considers the Italian economy as part of the euro area in order to properly take into account the role of the common monetary policy and the spillovers from (and to) the rest of the area.

The basic structure of the model is akin to the Global Economy Model (GEM) developed at the IMF.² There are monopolistic competition in the goods and labor markets, standard real and nominal frictions to match the persistence and inertia usually found in the data, an interest rate feedback rule for the monetary authority. Differently from other similar models, ours is rich in the terms of fiscal features, that allow to realistically analyze fiscal issues in a general equilibrium context. Fiscal policy is conducted at regional level. In each region we break down the Ricardian equivalence by introducing distortionary taxes on labor income, capital income and consumption, allowing for a realistic treatment of fiscal policy. On the expenditure side, we depart from the simplifying assumption that public expenditures are "pure waste". We carefully distinguish between different uses of public money. Specifically, we consider spending on final goods and services produced by the private sector, public employment and transfer to families. Decomposing public expenditures in its main components is important, as each one has different macroeconomic implications.³ In particular, we assume that

¹We received useful suggestions from Francesco Lippi, Alberto Locarno, Fabio Panetta, Morten Ravn, Stefano Siviero, Piero Tommasino, Aleh Tsyvinski, Joseph Zeira and from seminar participants at the Bank of Italy (November 2007), Computing in Economic and Finance annual meeting (Paris, June 2008), Society for Economic Dynamics annual meeting (Boston, July 2008), Italian Society of Public Finance annual meeting (Pavia, September 2008), Einaudi Institute for Economics and Finance (Roma, October 2008), LUISS University (Roma, October 2008), Dynare Conference (Oslo, August 2009) and Italian Ministry of Economic and Finance (Roma, December 2009). Usual disclaimers apply.

²See Pesenti (2008). See also Bayoumi (2004) for a non-technical description of the GEM. Several central banks have developed DSGE models for policy analysis. Among the others, the Fed has developed SIGMA (see Erceg et al. (2006)), the European Central Bank the New Euro Area Wide Model (see Coenen et al. (2007)).

 $^{^{3}}$ Rogerson (2007) argues that "it is essential to explicitly consider how the government spends tax revenues when assessing the effects of tax rates on aggregate hours of market work." For a formal analysis along these lines, see Leeper and Yang (2006).

public spending on private final goods is used as intermediate good and combined with public employment to produce public goods that positively affects the households' utility function. In this way, a trade-off between the welfare-enhancing public good and the misallocation of (goods and labor) resources induced by its production is introduced in the model.

We focus on scenarios where the public debt-to-annual gross domestic product (GDP from now on) is reduced from 105% to 95% over a five-year horizon. The scenarios differ in terms of tax rates and expenditure items that are changed to reach the target. The model parameters are calibrated to values commonly used in the literature and to replicate the great ratios of Italy and rest of the euro area. We assume that in the rest of the euro area lump-sum transfers are tuned in order to leave the public debt-to-GDP ratio unchanged.

We run simulations under perfect-foresight and assume that the only shocks perturbing the economy are the fiscal ones. We use Dynare to compute initial and final steady states and related transition path. We abstract from considerations related to lack of credibility, uncertainty, optimal Ramsey policy, the use of fiscal instrument to stabilize business cycle and to fiscal coordination issues between Italy and rest of the euro area.⁴

Along the transition nominal and real rigidities contribute, jointly with the gradual implementation of fiscal measures, to prolong the adjustment of the economy towards the new long run equilibrium. So we report long run (final steady state) and short-medium run (transition) macroeconomic domestic effects and spillovers to the rest of the euro area. We also provide a measure of the effects on welfare in terms of consumption equivalents. Finally, we perform sensitivity analysis to check for the robustness of results.

Results are as follows. First, we show that fiscal distortions are quantitatively relevant. For a given public debt-to-GDP ratio, tax rate cuts compensated by lower lump-sum transfers have clear welfare-improving implications. To the contrary, increases in expenditures (financed by lower lumpsum transfers) aimed at the provision of welfare-enhancing public goods, have negative welfare effects. The reason is that the increase in welfare related to the higher level of public good is more than compensated by the increase in economic distortions (on private goods and labor supply) associated to its production. Second, and consistently with the above results, the best way to accomplish a reduction in the public debt-to-GDP ratio is by lowering tax rates while, at the same time, reducing expenditures by more than would be needed with unchanged tax rates. In particular, a simultaneous reduction in public expenditures and tax rates that achieves the targeted reduction of the public debt has the highest long run steady-state expansionary effects on GDP and on all its components. The former increases by 5 to 7% of the initial steady state level, depending on the exact composition

⁴On the optimal Ramsey problem see Juillard and Pelgrin (2007).

of the adjustment. Moreover, among expenditures it is preferable to cut purchases of goods and services or public employment rather than transfers to households. Third, spillovers to the rest of the euro area are expansionary and sizeable (long run GDP in the rest of the euro area increases by 2-3%). Finally, on impact and along the transition GDP, investment and private consumption are stable or grow. When public purchases (a component of internal demand) or government employment (as GDP includes also the public sector wage bill) are being cut, GDP growth remain subdued along the transition.

Our findings are interesting along several dimensions. We contribute to the debate on the quantitative relevance of the macroeconomic effects of fiscal measures. In his Presidential Address to the AEA discussing the "Macroeconomic Priorities", R. Lucas (2003) argues that the welfare gains from supply side fiscal policies would be sizeable and equivalent to increases of about 5 to 15 percent in overall consumption levels. Also Feldstein (2008) discusses "how the effects of taxes on economic behavior are important for revenue estimation, for calculating efficiency effects, and for understanding short-term macroeconomic consequences." Mankiw and Weinzierl (2006) use standard growth models to assess the supply side effects of tax cuts and conclude that "in all models considered, the dynamic response of the economy to tax changes is too large to be ignored". They also show that the results obtained using the standard neoclassic growth model with infinitely lived agents - the framework considered in this paper - are robust to departures, like that of assuming agents with finite horizons or including a share of rule-of-thumb consumers.⁵

One of the results that we obtain is that there is a wide margin to reduce public expenditures with limited welfare costs. This conclusion supports those obtained by Afonso, Schuknecht and Tanzi (2005), although from a completely different perspective. Their study applies Data Envelope Analysis to assess the "efficiency frontier" of the public sector in the provision of public services and conclude that the same level of public services could be attained with 1/4 less public spending. This result is surprisingly close to what we find.

Our contribution is also related to the empirical literature on the non-Keynesian effects of fiscal policy.⁶ This literature has considered fiscal consolidations (variously defined) of OECD countries in order to obtain some indications on the characteristics that most likely would lead to successful (i.e. lasting) adjustments. The main conclusion are that (i) adjustments concentrated on the expenditure side of the budget more than on the revenue

⁵We extend the model to include non Ricardian (or rule-of-thumb) agents in the robustness analysis and confirm the findings of Mankiw and Weinzierl (2006).

⁶See, among the others, Alesina and Perotti (1995, 1997), Giavazzi and Pagano (1990, 1996), McDermott and Wescott (1996), Alesina and Ardagna (1998).

side and (ii) large adjustments (measured by the reduction in the debt-to-GDP ratio) tend to have more non-Keynesian effects. The main theoretical argument behind these results is that agents are forward looking and therefore any sustainable reduction in public expenditure would generate a wealth effect (agents foresee less taxes) leading to an increase in consumption, investment and economic activity. This wealth effect could – under certain circumstances (as in cases of very high debt-to-GDP ratio at the beginning of the consolidation phase) – dominate against the (Keynesian) direct depressing effect coming from cuts in public expenditures. Our general equilibrium model formalizes most of these channels and allows weighting them in a sound quantitative manner.

Other papers strongly related to ours are Coenen, McAdam and Straub (2007) and Coenen, Mohr and Straub (2006). In particular, the latter analyzes costs and benefits of fiscal consolidation scenarios in the euro area, using a less detailed description of fiscal policy that we use. Their results point to significant positive long-run effects on the main macroeconomic variables, mainly when the improvement in the budget position is used to lower distortionary taxes.

The paper is organized as follows. Section 2 provides a discussion of the setup of the model. Section 3 presents the results of the baseline fiscal consolidation scenarios. Section 4 discusses the transition dynamics of the different fiscal consolidation strategies, while section 5 provides robustness checks. Section 6 concludes.

2 The Model

In this section we initially illustrate the model setup, focusing mainly on the fiscal features. We then report the calibration and the model-based fiscal consolidation scenarios.

2.1 The Setup

There are two regions, Home and rest of the euro area, having different sizes and sharing the monetary policy and currency. In each region there are households and firms. Each household consumes a final good, which is a composite made of intermediate nontradable and tradable goods. The latter are produced domestically or imported. Households participate in financial markets and smooth consumption by trading a risk-free one-period nominal bond. They also own domestic firms and capital stock. The latter is rented to domestic firms in a perfectly competitive market. All households supply differentiated labor services to domestic firms and act as wage setters in monopolistically competitive labor markets by charging a markup over their marginal rate of substitution between consumption and leisure.

On the production side, there are perfectly competitive firms that produce the final goods and monopolistic firms that produce the intermediate goods. The two final goods (consumption and investment goods) are sold domestically and are produced combining all available intermediate goods using a constant-elasticity-of-substitution (CES) production function. Intermediate tradable and nontradable goods are produced combining domestic capital and labor, that are assumed to be mobile across sectors. Intermediate tradable goods can be sold domestically and abroad. Because intermediate goods are differentiated, firms have market power and restrict output to create excess profits. We also assume that markets for tradable goods are segmented, so that firms can set two different prices, one for each market. To capture the empirical persistence of the aggregate data and generate realistic dynamics, we include adjustment costs on real and nominal variables, ensuring that, in response to a shock, consumption and production react in a gradual way. On the real side, quadratic costs prolong the adjustment of the capital stock. On the nominal side, they make wages and prices sticky.⁷

In the following section we describe in detail the fiscal policy setup and the household problem. In the Appendix we laid down the rest of the model.

2.2 Fiscal policy

Fiscal policy is set at the regional level. The government budget constraint is: a = a

$$\left[\frac{B_{t+1}^g}{R_t} - B_t^g\right] = (1 + \tau_t^c) P_t C_t^g + W_t L_t^g + Tr_t - T_t \tag{1}$$

where $B_t^g \ge 0$ is nominal public debt. It is a one-period risk-free nominal bond issued in the euro area wide market that pays a gross nominal interest rate R_t controlled by the monetary authority of the currency union. The variable C_t^g represents government purchases of goods and services, $W_t L_t^g$ is compensation for public employees (W_t is the nominal wage, L_t^g is the total number of hours worked in the public sector), Tr_t are lump-sum transfers to households. We assume that C_t^g has the same composition as private consumption. Hence it is pre-multiplied by the private consumption price index P_t . Total government revenues T_t are given by the following identity:

$$T_t \equiv \tau_t^{\ell} W_t L_t + \tau_t^c \left[P_t C_t + P_t C_t^g \right] + \tau_t^k \left[R_t^k K_{t-1} + \Pi_t^P \right]$$
(2)

where the τs are tax rates on labor income (τ_t^{ℓ}) , capital income (τ_t^k) and consumption (τ_t^c) , L_t is total amount of hours worked (in the public sector, L_t^g , and in the private sector, L_t^p , that is $L_t = L_t^p + L_t^g$), R_t^k is the rental rate of existing physical capital stock K_{t-1} and Π_t^P stands for dividends from ownership of domestic monopolistic firms.

⁷See Rotemberg (1982)

The public sector combines labor, purchases of goods and services and a constant stock of public building and land, \overline{BL}_g to produce public goods Y^g (as health, education, security, justice, etc...) according to the following CES production function:

$$Y_t^g = \left[\left(1 - \gamma_{_Lg} - \gamma_{_Cg}\right)^{\frac{1}{\alpha_g}} \overline{BL}_g^{\frac{\alpha_g - 1}{\alpha_g}} + \gamma_{_Cg}^{\frac{1}{\alpha_g}} C_t^g^{\frac{\alpha_g - 1}{\alpha_g}} + \gamma_{_Lg}^{\frac{1}{\alpha_g}} L_t^g^{\frac{\alpha_g - 1}{\alpha_g}} \right]^{\frac{\alpha_g}{\alpha_g - 1}}$$

where $\alpha_g > 0$ measures the degree of substitutability between the three kinds of input and γ_{Lg} , γ_{Cg} are the weights of government employment and purchases of goods and services, respectively $(0 < \gamma_{Cg} < 1 \text{ and } 0 < \gamma_{Lg} < 1)$. Both C_t^g and L_t^g are exogenously given.

Given the presence of public employment, and consistently with common practice in the national accounts statistics, we include the public expenditure for wages in the definition of GDP:

$$GDP_t = C_t + p_t^I I_t + C_t^g + p_t^{EXP} EXP_t - p_t^{IMP} IMP_t + w_t L_t^g \qquad (3)$$

where p_t^I , p_t^{EXP} , p_t^{IMP} , w_t are prices of respectively investment, export and import and wage expressed in units of the domestic consumption bundle.

We assume that the government uses a fiscal rule defined on a single fiscal instrument to bring the public debt as a percent of domestic GDP, b > 0, in line with its target b^* . We consider alternative instruments among the three tax rates $(\tau_t^\ell, \tau_t^k, \tau_t^c)$ and the three expenditure items (C_t^g, L_t^g, Tr_t) , depending on the considered fiscal consolidation scenario.⁸ We assume the following fiscal rule:

$$\frac{i_t}{i_{t-1}} = \left(\frac{b_t}{b^*}\right)^{\phi_1} \left(\frac{b_t}{b_{t-1}}\right)^{\phi_2} \left(\frac{GDP_t}{GDP_{t-1}}\right)^{\phi_3} \tag{4}$$

where i_t is one of the six fiscal instruments considered. Parameters ϕ_1 , ϕ_2 and ϕ_3 are lower than zero when the rule is defined on an expenditure item calling for a reduction in expenditures whenever the debt level is above target and for a larger reduction whenever the dynamics of the debt is not converging and/or the GDP growth is positive. To the contrary, they are greater than zero when the rule is on tax rates. Overall, the fiscal setup of the model is able to take into account many implications of different tax and expenditure items. This is essential in order to understand the macroeconomic effects of fiscal consolidation scenarios.

⁸So the scenarios differ for the fiscal item that is exogenously changed by the fiscal authority and the one that is endogenously changed to stabilize the debt according to the fiscal rule.

2.3 Households

In each country there is a continuum of symmetric households. Home households are indexed by $j \in [0; s]$ and Foreign households by $j^* \in (s; 1]$.⁹ Households' preferences are additively separable in consumption and labor effort. Households receive utility from consuming and disutility from working L_t hours. The expected value of household j lifetime utility is given by:

$$E_0\left\{\sum_{t=0}^{\infty}\beta^t \left[\frac{\widetilde{C}_t(j)^{1-\sigma}}{(1-\sigma)} - \frac{\kappa}{\tau}L_t(j)^{\tau}\right]\right\}$$

where E_0 denotes the expectation conditional on information set at date 0, β is the discount factor ($0 < \beta < 1$), $1/\sigma$ is the elasticity of intertemporal substitution ($\sigma > 0$) and $1/(\tau - 1)$ is the labor Frisch elasticity ($\tau > 0$).

The consumption bundle $\widetilde{C}_{t}(j)$ is given by:

$$\widetilde{C}_t(j) = \left[\omega^{\frac{1}{\theta}} C_t(j)^{\frac{\theta-1}{\theta}} + (1-\omega)^{\frac{1}{\theta}} Y_t^{g\frac{\theta-1}{\theta}}\right]^{\frac{\theta}{\theta-1}}$$

where $\theta > 0$ measures the degree of substitutability between private (C) and public goods (Y^g) while $0 \le \omega \le 1$ is the weight of the private good in the consumption bundle. When $\omega = 1$, the level of the public good does not alter private consumption decisions.

The budget constraint of agent j is:

$$\frac{B_t(j)}{(1+R_t)\mu_t} - B_{t-1}(j) \leq (1-\tau_t^k) \left(\Pi_t^P(j) + R_t^K K_{t-1}(j) \right) + \\
+ (1-\tau_t^\ell) W_t(j) L_t(j) - (1+\tau_t^c) P_t C_t(j) - P_t^I I_t(j) \\
+ Tr_t(j) - AC_t^W(j)$$

Home agents hold a one-period risk-free bond, B_t , denominated in the currency of the monetary union. The short-term nominal rates R_t is paid at the beginning of period t and is known at time t. It is directly controlled by the monetary authority. A financial friction μ_t is introduced to guarantee that net asset positions follow a stationary process and the economy converge to a steady state.¹⁰ We assume that government and private bonds can be traded internationally in the same market. Households own all domestic firms and there is no international trade in claims on firms' profits. The variable Π_t^P includes profits accruing to the Home household. We assume that profits are equally shared across households. The variable I_t is

⁹The population of the monetary union is normalized to one. The parameter s is the size of the Home population, which is also equal to the number of firms in each Home sector (final nontradable, intermediate tradable and intermediate nontradable). Similar assumptions holds for 1 - s in the rest of the euro area.

¹⁰Revenue from financial intermediation are rebated in a lump-sum way to agents in the rest of euro area. See Benigno (2009).

investment bundle in physical capital and P_t^I the related price index, which is different from the price index of consumption because the two bundles have different composition.¹¹ Home agents accumulate physical capital K_t and rent it to domestic firms at the nominal rate R_t^k . The law of motion of capital accumulation is:

$$K_t(j) = (1 - \delta) K_{t-1}(j) + (1 - AC_t^I(j)) I_t(j)$$

where δ is the depreciation rate. Adjustment cost on investment AC_t^I is :

$$AC_{t}^{I}(j) = \frac{\phi_{I}}{2} \left(\frac{I_{t}(j)}{I_{t-1}(j)} - \delta \right)^{2}, \ \phi_{I} > 0$$

Finally, Home households act as wage setters in a monopolistic competitive labor market. Each household j set her nominal wages taking into account of labor demand and adjustment costs AC_t^W on the nominal wage $W_t(j)$:

$$AC_{t}^{W}(j) = \frac{\kappa_{W}}{2} \left(\frac{W_{t}(j)}{W_{t-1}(j)} - 1\right)^{2} W_{t}L_{t}, \ \kappa_{W} > 0$$

The costs are proportional to the per-capita wage bill of the overall economy, $W_t L_t$. Similar relations hold in the Foreign country, with the exception of the intermediation frictions in the financial market.

2.4 Calibration

The model is calibrated at a quarterly frequency. We set some parameter values so that steady-state ratios are consistent with 2007 national account data, which are the most recent and complete available data. We choose to not use projection for 2010 public debt-to-GDP ratios, even if available, for two reasons. First, remaining data on fiscal variables are not available yet or are surrounded by a high degree of uncertainty. Second, given that it is likely that 2010 public expenditure and debt are higher than the current levels, our estimates represent a lower bound of the effects, so that we put ourselves on the conservative side. For remaining parameters we resort to previous studies and estimates available in the literature.¹²

Table 1 contains parameters that regulate preferences and technology. Parameters with a "*" are related to the rest of the euro area. We assume that discount rates and elasticities of substitution have the same value across the two regions. The discount factor β is set to 0.9875, so that the steady state real interest rate is equal to 5 per cent on an annual basis. The value for the intertemporal elasticity of substitution, $1/\sigma$, is 1. The Frisch labor

¹¹See the Appendix for more details.

¹²Among others, see Forni, Gerali and Pisani (2009a) and Forni, Monteforte and Sessa (2009).

elasticity is set to 2. The weight of the private good ω in the utility function is $0.8.^{13}$ The elasticity of substitution between private and public goods, θ , is set at $1.5.^{14}$ The depreciation rate of capital δ is set to 0.025 on a quarterly basis.

In the production functions of tradables the elasticity of substitution between labor and capital is set respectively to 0.85 in Italy and 0.9 in the rest of the area. In the production functions of non-tradables to 0.79 and 0.95. The biases towards private capital is set to 0.75 and 0.7 in the Italian and rest of the area tradable sectors, respectively; to 0.7 for both areas in the non tradable sectors. In the production function of the public sector the elasticity of substitution between inputs (labor, fixed stock of public capital and intermediate goods) α_g is equal to 0.79, the biases towards intermediate goods γ_{Cg} and labor γ_{Lg} are set to 0.15.

In the final consumption and investment goods the elasticity of substitution between domestic and imported tradable is set to 1.5, while the elasticity of substitution between tradables and non tradables to 0.5. The bias for the composite tradeable to 0.55 in Italy and 0.5 in the rest of the area. The biases for the domestically produced and that for the composite tradable goods are chosen to match the Italy-Euro area import and exportto-GDP ratios. The population size of Italy, n, is set to 0.2 (we normalize the whole Euro area population to 1).

Table 2 reports gross markups in the tradable, non-tradable and labor markets. They are all set to 1.2. Hence the net markups are set to 20%. We obtain this number by calibrating the sector-specific elasticities of substitution between varieties to 6.1^{5}

Table 3 contains parameters that regulate the dynamics. Adjustment costs on investment change are set to 1. Both nominal wage and price quadratic adjustment costs are set to 60, which corresponds to an average frequency of wage and price adjustment roughly equal to 4 quarters. The two parameters regulating the adjustment cost paid by the Italian private agents on their net financial position are set to 0.01.

Parametrization of systematic feedback rule followed by the fiscal and monetary authorities are reported in Table 4. In the fiscal policy rule (4) we set $\phi_1 = 1.5$ and $\phi_2 = \phi_3 = 15$ for both Italy and rest of the area. The chosen values allow reach the public debt target in more or less eight years in all the simulations. Their sign is positive when the fiscal instrument in the

¹³There is not clear empirical evidence that we can use in the calibration of this parameter. We check the robustness of the results in section 5.

¹⁴In the robustness section we will discuss also the results when the elasticity of substitution is lower (we will assume $\theta = 0.8$). Most contributions assume that private and public consumption are substitutes. For example, Prescott (2002) assumes they are perfect substitutes.

¹⁵For an analysis of the macroeconomic effects of different degree of murkups in a model similar to the one used in this paper, see Forni, Gerali and Pisani (2009b).

rule is a tax rate, it is negative when the instrument is a public expenditure. The central bank of the Euro area targets the contemporaneous Euro-area wide consumer price inflation (the corresponding parameter is set to 1.7) and the output growth (the parameter is set to 0.4).¹⁶ Interest rate is set in an inertial way and hence its previous-period value enters the rule with a weight equal to 0.9.

Table 5 reports model-based and actual steady-state great ratios and tax rates under our baseline calibration. Private consumption, bilateral imports and exports match the data rather well, while private investment in Italy is somehow underestimated. On top of that, consistently with available data, we assume a zero steady state net foreign asset position for the Italian economy: this implies that - in steady state - the *net* financial position of the Italian private sector equals the level of the Italian public debt. This assumption holds in both the initial and final steady state (but not along the transition).¹⁷

As for fiscal policy variables, it must be noted that some expenditure items (as purchases C^g as a ratio to GDP) are perfectly matched as they are exogenous. For other items, as the public wage bill and the interest expenditure, we calibrate the share of public employees over the total number of employees and the level of public debt-to-annual GDP ratio to replicate the actual data. As the wage and interest rates are endogenous, however, we don't match exactly the corresponding expenditure components. Tax rates are calibrated using effective average tax rates estimates for 2007 taken from Eurostat (2008). The tax rate on wage income τ^{ℓ} is set to 43.1 per cent in Italy and to 38.7 in the rest of the Euro area. The tax rate on physical capital income τ^k to 29.0 and 30.1, while the tax rate on consumption τ^c to 16.9 in Italy and to 19.2 in the rest of the euro area. The public debt-to-yearly GDP ratio is calibrated to 105 for Italy and to 60 for the rest of the euro area.

3 Results

In what follows we simulate the model to quantitatively assess the macroeconomic effects of several fiscal consolidation strategies implemented in the

¹⁶The euro area-wide consumer price inflation rate and GDP are weighted (by the regional size) geometric average of the corresponding regional variables.

¹⁷We have done robustness analysis assuming steady state Italian net financial position different from zero in the initial steady state and a value different from zero in the final steady state. Results, available from the authors upon request, are not greatly affected. This is to be expected. Because we have a monetary union framework, we don't have a flexible nominal exchange rate that induces "valuation effects" on the financial position through its fluctuations.

Italian economy.¹⁸

We initially assess the optimal composition of the budget (in terms of expenditures and revenues) for given level of Home public debt-to-GDP ratio (section 3.1). We proceed in two related steps. First, we show that reductions in tax rates or expenditure items can have significant welfare gains. Second we simulate a simultaneous cut in tax rates and expenditure items such that the debt level remains unchanged and compute the level of tax rates and expenditure items that maximizes the welfare level. For simplicity, we focus on steady state comparisons.

In section 3.2 we present the main results of the paper, those regarding fiscal consolidation. We permanently reduce the Home public debt-to-annual GDP ratio by 10 percentage points over a five-year horizon. The reduction can be obtained through adjustments in revenues and expenditures by appropriately changing the fiscal instrument in the fiscal rule (4). We show the long run (steady-state) and dynamic (transitional) macroeconomic and welfare impact of the possible alternative fiscal consolidation strategies.

The main result is that reductions of fiscal distortions have sizeable expansionary effects on the Home economy and positive effects on Home welfare. In particular, fiscal consolidations based on simultaneous reductions of tax rates and expenditure items can have positive effects on activity and welfare both in the long and short run.

3.1 Optimal expenditure and revenue composition for given level of public debt

To show the quantitative relevance of tax and expenditure distortions, that is how much we can improve welfare if we reduce them, we simulate under perfect foresight the effects of compensated reductions in the level of distortionary taxation and government expenditures. These exercises also help in understanding the transmission mechanism of the model and the results of the consolidation scenarios, reported in the next section. The reductions in *tax distortions* are achieved via reductions in tax rates compensated by

¹⁸The results that we present with respect to the Italian economy mosltly hold also for other euro area countries. In a companion paper (Forni, Gerali and Pisani, 2009) we simulate the model alternatively calibrated to Germany and Belgium. The analysis of Germany and Belgium provides enough cases to assess the situation of most euro area countries. As argued in the paper, in fact, euro area countries are relatively homogenous in terms of expenditure shares of GDP and also in terms of fiscal variables (as the total level of expenditures and revenues). They mainly differ in size, in the degree of openness and in the level of public debt. In the paper we provide robustness exercises along several dimensions (size and openness of the country among the others) in order to show how one can adjust the basic insights gained from the German and Belgian cases to the rest of euro area countries. For public debt, simulations show that the initial level of public debt-to-GDP ratio does not greatly affect the main results, that mainly depend upon the initial level of distortionary public expenditure and taxation.

reductions in lump-sum transfers. The reduction in *expenditure distortions* is obtained reducing C^g and L^g while at the same time increasing transfers. Remember that in our setup not only tax rates, but also expenditures are distortionary as they change the optimal allocations of private agents (through both the wealth effect and the public goods in the utility function).

Table 6 shows the percentage changes with respect to the initial steady state levels for the main macroeconomic variables. We report also the percent change in welfare between initial and final steady state. The measure is expressed in terms of consumption equivalents, that is the constant percentage change in consumption level (\tilde{C}) that would deliver the same utility as the one achieved in the scenario under consideration. The measure does not take into account the welfare effects during the transition, that are illustrated in the next section.

The first three columns of the table show the long-run effects of reducing transfers to households (Tr) by 1 per cent of GDP and exactly compensating this expenditure reduction with tax rates reductions (either on labor income, capital income or consumption) as to leave the level of public debt as a ratio to GDP unchanged. Since transfers are in the model equivalent to a negative lump-sum tax, this procedure delivers a reduction in tax rates leaving unchanged the total amount of net taxes (that is taxes minus transfers, as a percentage of GDP) that agents have to pay.

The table shows that the reduction in tax rates, compensate by lower lump-sum transfers, produces an increase in welfare between 0.5 and 1.2 per cent. The reduction in labor income tax rate (column 1) induces a decrease in real wages (w) while at the same time a substantial increase in aftertax real wages $((1 - \tau^{\ell})w)$, employment and consumption. The increase in employment brings about also an increase in investment. Similarly, the cut in consumption tax rate (column 3) leads to a reduction in real wages and to an increase in employment. At the same time it favors consumption over investment and therefore limits capital accumulation. In sum, a cut in this tax rate leads to a limited increase in employment, investment, output and welfare. Also, cuts to the consumption tax rate apply to both domestically produced and imported goods, while cuts to labor income or capital income taxes reduce the cost of production only of domestically produced goods. Finally, in the case of a reduction in the capital income tax rate (column 2), the increase in investment drives up output, while consumption is subdued as the reduction in capital income taxes makes it relatively more costly.¹⁹

As the welfare and efficiency gains related to cuts in consumption tax

¹⁹The size of the welfare gains are rather robust to alternative calibrations. In particular, we have done some robustness check with respect to the parameters of the production function (as the elasticity of substitution between labor and capital) and utility function (as the intertemporal elasticity of substitution and the level of the disutility of the working effort) and there are not substantial changes in the results.

rates tend to be significantly smaller than those due to cuts in labor and capital income rates, the analysis in the rest of the paper will focus on those two latter rates. It must be kept in mind, however, that consumption taxes are still in the model (although fixed) and contribute to the calibration of steady state values.

The columns 4 and 5 of the table show the effects of reducing *expenditure distortions*. This is achieved by increasing lump-sum transfers by 1 per cent of GDP while at the same time reducing by the same amount government purchases (column 4) or public employment (column 5). As the increase in transfers corresponds to a reduction in net taxes, without reductions in tax rates, the move achieves a reduction in the overall level of taxation without changing tax rates. On the one hand, welfare improves due to the positive income effect; on the other, the reduction in the provision of the utility-enhancing public good has a negative effect on welfare. Overall, both in column 4 and 5 the welfare gain is positive, although tend to be smaller than the gain obtained by reducing labor and income tax rates. GDP decreases, mainly because of the reduction in its public component (both purchases of goods and services or the public wage bill are part of GDP; see equation 3).

Up to this point we have analyzed the gains in implementing compensated tax rates and expenditure cuts. We now assess the trade-off existing when the reduction in tax rates is achieved through the reduction in welfare-improving public expenditures. That is, the cuts in tax rates are compensated not by lump-sum transfers but via reduction in purchases C^g or public employment L^g , that are used to produce the public good.

In Figure 1 we report the welfare level in steady state for different combinations of labor and capital income taxes, while setting all other parameters at their baseline values. The figure plots the welfare level assuming that the reduction in tax rates is compensated by cuts in one of the three expenditure items (purchases C^{g} , public employment L^{g} and transfers Tr) in order to leave the public debt-to-GDP ratio unchanged. The point in the figure labelled *initial steady state* has a welfare level normalized to 1 (in the initial steady state $\tau^{\ell} = 0.431$ and $\tau^{k} = 0.29$). The picture shows that reducing one or both rates increases the welfare level, regardless of the expenditure item that is being reduced. Welfare increases almost linearly when the reduction in tax rates is compensated by cuts in transfers, as the change simply reduces tax distortions. When the expenditure reduction is concentrated on C^{g} , the welfare increases up to a maximum of about 3%. At the maximum τ^{ℓ} is at about 29% and τ^{k} at about 23%. This implies a cut in the former of about 14 points and in the latter of 6 points. When it is concentrated on L^{g} , welfare goes up to a maximum of about 2% (with τ^{ℓ} at 29% and τ^{k} at 25%). In both cases total expenditure in Italy would decrease by about 1/4, roughly the same number that Afonso, Schuknecht and Tanzi (2005) find, using a completely different approach.

To sum up, based on our calibration, tax and expenditure distortions seem to be significant. Moreover, there is a wide margin to cut tax rates and expenditures while increasing the level of welfare. In particular, our results suggest that welfare would increase for simultaneous cuts in the labor and capital income tax rates, compensating the revenue loss by reducing public expenditures.

3.2 The long-run effects of the fiscal consolidation

We now consider scenarios where the target level of debt-to-GDP ratio is permanently reduced by 10 percentage points over five years. The size of this reduction is realistic, although rather ambitious.

We consider fully credible and fully anticipated consolidation plans and run perfect-foresight simulations. In this section we compare steady states before and after the consolidation, while in the next one we study the adjustment path of endogenous variables towards the new steady state level.

Table 7 reports steady state results. The first two columns - labelled $(B, \tau^{\ell}), (B, \tau^{k})$ - assume that the consolidation is achieved increasing along the transition one tax rate at a time (on labor income and capital income, respectively) following the fiscal rule (4), leaving public expenditure for goods and services (as ratio to GDP) and for employment (as ratio to total employment) unchanged.²⁰

In the next three columns of Table 7 - labelled (B, C^g) , (B, L^g) and (B, Tr) - the consolidation is achieved imposing along the transition the fiscal rule defined on one expenditure item at a time (purchases of goods and services, public employment and transfers, respectively), leaving tax rates unchanged. The columns after the fifth consider scenarios where, in order to reduce the Home public debt-to-GDP ratio to the target, tax rates are exogenously reduced by five percentage points and one expenditure item at a time is endogenously reduced through the fiscal rule. By reducing both tax rates by 5 percentage points, total primary expenditures have to be cut by about 4% of GDP, quite a significant amount.²¹

The intuition behind the steady-state results is as follows. In the scenarios of tax-based consolidation, tax rates are increased along the transition. Once the debt target is achieved and interest expenditure on public debt is

²⁰Results are only slightly different if we assume that expenditures remain unchanged in real terms, instead of as a percentage of GDP. Since GDP increases for all three tax cuts, fixed expenditures in real term would imply that they would decrease in terms of GDP. Therefore, the positive effects (on the macro variables and on steady state welfare) would be larger. As expenditures, especially in Italy, tend to grow with GDP, we feel more confident with our baseline assumption.

 $^{^{21}}$ We could have considered larger tax cuts. These, however, would have implied reductions in total primary expenditures larger than 4% of GDP, an amount difficult to achieve in the horizon that we consider for the transition.

reduced, tax rates can stabilize at a final steady-state level lower than the initial one. Similarly, in the scenarios of public expenditure-based consolidation, public expenditures are cut along the transition but eventually end up to a final steady-state level higher than the initial one, substituting for the lower interest outlays. Lastly, reducing both expenditures and taxes along the transition implies that the lower steady-state interest rate payment is divided between lower expenditures and taxes.

The first two columns of Table 7 shows that reducing tax rates induces an increase in output, which is slightly stronger for lower labor income tax rate. In the latter case there is a positive reaction in hours worked, that induces higher consumption (households substitute consumption for leisure) and investment (capital is more productive when employment is higher). In the case of lower capital income tax, investment strongly increases while the increase in consumption and employment is relatively low.

Columns 3-5 show the effects of higher steady state public expenditure for goods, employment and lump-sum transfers. The latter have zero effect, given that the net financial asset position of the Italian economy (equal to the sum of private and public sector asset positions) is equal in both the initial and final steady state and change in transfers do not affect households' first order conditions. In the other two cases, output increases by the same amount, albeit for different reasons. Higher public expenditure for goods and services induces a decrease in private demand for consumption and an increase in supply driven by employment and capital (higher investment). Higher public expenditure for employment induce an increase in the wage component of output (see equation 3), while private demand decreases.

Columns 6-8 report the results assuming a reduction in labor income taxes equal to 5 percentage points. Output increases less when public consumption and employment are reduced, because, differently from lump-sum transfers, they directly affect the GDP. To the contrary, private consumption increases more, as more resources are made available for private (households and firms) demand.

A similar picture emerges from columns 9-11. Similarly to the previously considered scenarios, in the new steady state both capital income taxes and public expenditures are reduced. Also in this case, the lower increase in GDP and the higher increase in private consumption is associated to the C^g and L^g scenarios.

A similar ranking and logic apply when both taxes are simultaneously reduced (columns 12-14). There are expansionary effects on the economic activity, that are roughly equal to the sum of effects obtained when tax reductions are implemented separately.

We conclude that all tax-based reforms have positive effects on the steady state welfare, which increases with respect to the initial one. The biggest effect is obtained when all taxes and expenditures are reduced. This means that utility provided by the public good is more than compensated by the distortions associated to taxation, public employment and purchases. Consistently with this statement, the steady state welfare deteriorates in the scenarios reported in columns 3 and 4, when tax rates are not changed and public expenditures increase in the steady state.

Finally, spillovers to the rest of the euro area are significant. The effects on the rest of the euro area, relative to the domestic ones, are slightly bigger than the relative size of the country. They are generally positive, given that the expansionary effects of reforms on the Home supply side imply higher Home imports and cheaper Home goods for all households in the area. Consistently, the Home terms of trade - defined as the price of Home imports to the price of Home exports (both expressed in terms of Home consumption units) - deteriorate, while the Home real exchange rate, defined as the ratio of rest of euro area to Home consumer prices, depreciates.

Overall, the main result is that in the euro area country specific fiscal consolidation strategies that reduce taxes and public expenditures have longrun expansionary effects on the domestic production and hence on economic activity and welfare as well as positive spillovers on the rest of the euro area.

4 Transition dynamics

In the previous section we have seen that the permanent reduction in the public debt-to-GDP ratio is compatible with a significant long run steadystate increase in economic activity and welfare gains when steady state expenditures and revenues are reduced at the same time. In this section we analyze the related transition from the initial steady state to the final one. After a permanent fiscal shock, the economy does not jump immediately from one steady state to the other, because (a) the shock is implemented in a gradual manner and (b) presence of nominal and real rigidities (nominal sticky prices and wages, adjustment costs on investment) slows the adjustment process.

In the following we focus on scenarios where - over a five-year horizon - the target level of the debt-to-GDP ratio permanently decreases by 10 percentage points and both labor and capital income tax rates are cut by 5 percentage points. As shown in the previous section, this policy strategy induces the higher increase in the long-run steady state welfare (columns 12-14 in Table 7). As usual, we consider three scenarios. The first (scenario C^g) corresponds to the case where the cut falls on public expenditure for intermediate goods. In the second (scenario L^g) the cut falls on the expenditure for public employment. Finally, the scenario Tr is characterized by a reduction in lump-sum transfer to households. Each expenditure item is adjusted according to the fiscal rule (4).

Figure 2 shows the path of the main fiscal variables and of GDP, while Figure 3 the path of the remaining main macroeconomic variables. Figure 2 shows that the path of public debt is similar across scenarios. It slowly converges to the target in about 8 years. Also the GDP show a similar path across scenarios. It is always above the baseline and increases gradually over time. The other macroeconomic variables display a somehow different pattern depending on the type of fiscal consolidation considered (Figure 3). In the scenario C^g there is a strong increase in consumption of private goods on impact, driven by the amount of resources made available by the lower public good. The Home inflation rate increases, contributing to lower the domestic real interest rate (not reported). The latter decreases because the increase in domestic inflation is not compensated by an increase in the euro area wide nominal interest rate. As employment increases and the supply of goods expands, compensating for the increase in aggregate demand, the inflation rate moderates and consumption slows. In the medium run consumption persistently increases, as tax distortions and public expenditures are reduced. In the other two scenarios consumption does not increase on impact, as the cut in transfers or the public wage bill reduces households disposable income and therefore moderates initially the increase in private consumption.

The described macroeconomic paths have a positive effect on welfare. Table 8 reports a measure of welfare along the transition path and in the final steady state. It is measured in terms of consumption equivalents, that is the constant change, x, in initial steady state (ss) consumption that induces the same discounted flow of utility as the actual one, that is:

x s.t.
$$\sum_{i=1}^{\infty} \beta^{i} U\left(xC_{ss,L_{ss}}\right) = \sum_{i=1}^{\infty} \beta^{i} U\left(C_{i,L_{i}}\right)$$

According to our results, consolidations based on simultaneous reductions in tax rates and public expenditures on employment and purchases of goods and services produce the highest increase in welfare, due to the strongest wealth effect associated to the reduction in fiscal distortions.

5 Robustness

We performed robustness checks on three important aspects of the model. First, with respect to the elasticity of labor supply, which drives the response of employment to tax cuts. Second, with respect to the role of the public good in the utility function, changing its weight (ω) and its degree of complementarity/substitutability with the private one (θ). Third, we introduce non Ricardian (or rule-of-thumb) agents, i.e. agents that do not participate in the financial markets and consume their current (labor plus government transfers) net income. The latter two robustness exercises are meant to increase the negative welfare effects of cutting expenditures and see whether, for realistic alternative calibration of these parameters, our main results (in particular, that the positive effects due to tax cuts more than compensate the negative effects coming from expenditures cuts) can be overturned.

The first three columns of Table 9 report our baseline scenario (same as in the last three columns of Table 7). The columns from forth to sixth assume $\tau = 3$, thus a Frisch labor elasticity of 0.5 (instead of 2 as in the baseline scenario), a rather extreme value given that most estimated DSGE models place this elasticity in a range between 1 and 2. Results are somehow expected: employment increases by less, leading to a lower increase in investment, consumption and output.

The columns (7)-(9) replicate the baseline scenario assuming $\omega = 0.5$ (instead of 0.8), thus giving a weight equal to one half to the public good in the consumption bundle. In this case we observe a drop in the welfare gains of the fiscal consolidation, consistently with the fact that it requires cuts in expenditures. The drop is higher especially for cuts to public employment and purchases, as these expenditure items affect directly the production of the public good, while is much more limited for cuts to transfers. It must be noted, in any case, that welfare gains remain in general positive and significant. As for the effects on the macro variables, since public and private goods are substitutes (in the baseline we assume $\theta = 1.5$), the drop in the public good leads to a slightly higher increase in private consumption.

In the next three columns, (10)-(12), we assume that public and private goods are complements ($\theta = 0.8$). This implies that reductions in purchases or public employment (that reduce the provision of the public good) decrease the marginal utility of private consumption. No surprise therefore that in this scenario private consumption increases by less, although moderately.

Finally we evaluate the robustness of our results with respect to the introduction of a share of non Ricardian agents (NR), λ , equal to 35 per cent. Non Ricardian agents are assumed to consume their current disposable income, that is:

$$(1 + \tau_t^c) P_t C_t^{NR}(j) = (1 - \tau_t^\ell) W_t(j) L_t^{NR}(j) + Tr_t(j)$$

The results are shown in columns (13)-(15) and are only slightly different from the baseline. This is in line with the finding of Mankiw and Weinzierl (2006), among others. The reason is that non Ricardian agents do not smooth consumption and therefore do not contribute to pin down the steady state level of the capital stock. Moreover since, for simplicity, we assume that non Ricardians supply the same quantity of hours as Ricardian agents, they do not contribute to the choice of employment either.

Overall these robustness checks broadly confirm our baseline results. In particular in all cases we obtain that reductions in the debt to GDP ratio obtained via a concomitant reduction in expenditures and revenues is welfare improving. In general, the consequences of the different assumptions on the parameter values that we have considered are rather limited, both on the macroeconomic variables and on the welfare levels.

6 Concluding remarks

We have assessed the macroeconomic and welfare implications of different fiscal consolidation scenarios using a medium scale two-areas dynamic general equilibrium currency-union model – calibrated to replicate the main Italian and euro area macroeconomic and fiscal policy aggregates. The scenarios that we have considered envisage a decreases in the public debt of 10 percentage points of GDP in 5 years. On the basis of our simulations, a significant debt to GDP decrease obtained via reducing both expenditure and taxes can be welfare improving. The order of magnitude of these welfare gains is comparable with those suggested by Lucas (2003).

Our simulations also suggest that simultaneous reductions in public expenditures and tax rates - that achieve the targeted reduction of the public debt - can have long run steady-state expansionary effects on GDP and on all its component. The former can increase by 5% to 7% of the initial steady state level, depending on the exact composition of the adjustment. Moreover, among expenditures it is preferable to cut purchases of goods and services or public employment rather than transfers to households. The spillovers to the rest of the euro area are expansionary and sizeable (long run GDP in the rest of the euro area would increase by 2-3%). Finally, along the transition GDP, private consumption and investment do not fall. These results are robust to alternative calibrations.

References

- Afonso, A, Schuknecht L. and V. Tanzi (2005), "Public Sector Efficiency: an International Comparison", *Public Choice* 123 (3-4).
- [2] Alesina, A. and S. Ardagna (1998), "Tales of Fiscal Adjustment", *Economic Policy*, No. 27.
- [3] Alesina, A. and R. Perotti (1995), "Fiscal Expansions and Adjustments in OECD Countries", *Economic Policy*, No. 21.
- [4] Alesina, A. and R. Perotti (1997), "Fiscal Adjustment in OECD Countries: Composition and Macroeconomic Effects", *IMF Staff Paper*, Vol. 44, No. 2.
- [5] Bayoumi, T. (2004), "GEM: A New International Macroeconomic Model", *IMF Occasional Paper* No. 239.
- [6] Bayoumi, T., D. Laxton and P. Pesenti (2004), "Benefits and spillovers of greater competition in Europe: a macroeconomic assessment", *NBER Working Paper* No. 10416.
- [7] Benigno P. (2009): "Price Stability with Imperfect Financial Integration", Journal of Money, Credit and Banking, Vol. 41, No. 1.
- [8] Coenen G., P. McAdam and R. Straub (2007): "Tax reform and labourmarket performance in the euro area: a simulation-based analysis using the New Area-Wide Model", *Journal of Economic Dynamics and Control.*
- [9] Coenen G., M. Mohr, R. Straub (2006): "Fiscal consolidation in the euro area: long-run benefits and short-run costs", *ECB Working Paper*.
- [10] Ergeg, C., C. Gust and L. Guerrieri (2006), "SIGMA: A New Open Economy Model for Policy Analysis", *International Journal of Central Banking* 2, 1-50.
- [11] Eurostat (2008): "Taxation trend in the European Union".
- [12] Feldstein M. (2008): "Effects of taxes on economic behavior", NBER Working Paper 13745.
- [13] Forni L., A. Gerali, M. Pisani (2009a): "The macroeconomics of fiscal consolidations in Euro area countries, *mimeo*, Banca d'Italia.
- [14] Forni L., A. Gerali, M. Pisani (2009b): "Macroeconomic effects of greater competition in the service sector: the case of Italy", *Macro*economic Dynamics, forthcoming.

- [15] Forni L., L. Monteforte, L. Sessa (2009): "The general equilibrium effects of fiscal policy: Estimates for the euro area", *Journal of Public Economics* 93 559-585.
- [16] Giavazzi, F. and M. Pagano (1990), "Can Severe Fiscal Contractions be Expansionary? Tales of Two Small European Countries", in O. J. Blanchard and S. Fischer (eds.), NBER Macroeconomics Annual, Cambridge, Massachusetts, MIT Press.
- [17] Giavazzi, F. and M. Pagano (1996), "Non-Keynesian Effects of Fiscal Policy Changes: International Evidence and Swedish Experience", *Swedish Economic Policy Review*, Vol. 3, No. 1.
- [18] Juillard M. and F. Pelgrin (2007): "Computing Optimal Policy in a Timeless Perspective: an Application to a Small-Open Economy", *Bank* of Canada Working Paper 2007-32.
- [19] Mankiw N.G. and M. Weinzierl (2006): "Dynamic scoring: a back-ofthe-envelope guide", Journal of Public Economics, Vol. 90.
- [20] Leeper E.M. and S. S. Yang (2006): "Dynamic Scoring: Alternative Financing Schemes", mimeo.
- [21] Lucas, R. E. (2003): "Macroeconomic priorities", American Economic Association Presidential Address.
- [22] McDermott, C. J. and R. F. Wescott (1996), "An Empirical Analysis of Fiscal Adjustment", *IMF Staff Paper*, Vol. 43, No. 4.
- [23] Pesenti, P. (2008), "The Global Economy Model (GEM): Theoretical framework", *IMF Staff Papers*, Vol. 55, No. 2.
- [24] Prescott, E.C. (2002): "Prosperity and depression", American Economic Review, Vol. 92 no.2.
- [25] Rogerson R. (2007): "Taxation and market work: is Scandinavia an outlier?", NBER Working Paper no. 12890.
- [26] Rotemberg, Julio J. (1982), "Monopolistic price adjustment and aggregate output. *Review of Economic Studies* 49, 517-31.

Appendix

In this Appendix we report a detailed description of the model, excluding the fiscal policy part and the description of the Households optimization problem that are reported in the main text.

A The setup

There are two regions, Italy and rest of the euro area, having different sizes and sharing the currency and the central bank. In each region there are households and firms. Each household consumes a final composite good made of non-tradable, domestic tradable and imported intermediate goods from the rest of the area. Households have access to financial markets and smooth consumption by trading a risk-free one-period nominal bond. They also own domestic firms and capital stock, which is rent to domestic firms in a perfectly competitive market. Households supply differentiated labor services to domestic firms and act as wage setters in monopolistically competitive markets by charging a markup over their marginal rate of substitution.

On the production side, there are perfectly competitive firms that produce the final goods and monopolistic firms that produce the intermediate goods. The three final goods (a private consumption, a private investment and a public consumption good) are produced combining all available intermediate goods in a constant-elasticity-of-substitution matter. Tradable and non-tradable intermediate goods are produced combining capital and labor in the same way. Tradable intermediate goods are split in domesticallyconsumed and export goods. Because intermediate goods are differentiated, firms have market power and restrict output to create excess profits. We assume that Italy and the rest of the euro area are segmented markets and the law of one price for tradables does not hold. Hence, each firm producing a tradable good sets two prices, one for the domestic market and the other for the export market. Since the firm faces the same marginal costs regardless of the scale of production in each market, the different price-setting problems are independent of each other.

To capture the empirical persistence of the aggregate data and generate realistic dynamics, we include adjustment costs on real and nominal variables, ensuring that, in response to a shock, consumption and production do not immediately jump to a new long-term equilibrium. On the real side, quadratic costs prolong the adjustment of the capital stock. On the nominal side, quadratic cost make wage and prices sticky.

Imperfect competition in product and labor markets is reflected in markups over marginal costs. The elasticity of substitution between products of different firms determines the market power of each profit-maximizing firm. The setup in the labor market is similar. Each worker offers a differentiated kind of labor services that is an imperfect substitute for services offered by other workers. The lower the degree of substitutability, for example because of skill differences or anti-competitive regulation, the higher is the markup and the lower employment in terms of hours. Hence, markups are modeled by a single parameter.

A The model

In what follows we illustrate the Home economy (Italy). The structure of the Foreign economy (the rest of the euro area) is similar and to save on space we do not report it.

A Final consumption and investment goods

There is continuum of symmetric Home firms producing Home final nontradable consumption under perfect competition. Each firm producing the consumption good is indexed by $x \in (0, s]$, where the parameter 0 < s < 1is a measure of country size. Foreign firms producing the Foreign final consumption goods are indexed by by $x^* \in (s, 1]$ (the size of the monetary union is normalized to 1). The CES production technology used by firm xis:

$$A_{t}(x) \equiv \left(\begin{array}{c} a_{T}^{\frac{1}{\phi_{A}}} \left(a_{H}^{\frac{1}{\rho_{A}}} Q_{HA,t}(x)^{\frac{\rho_{A}-1}{\rho_{A}}} + (1-a_{H})^{\frac{1}{\rho_{A}}} Q_{FA,t}(x)^{\frac{\rho_{A}-1}{\rho_{A}}} \right)^{\frac{\rho_{A}}{\rho_{A}-1}\frac{\phi_{A}-1}{\phi_{A}}} \\ + (1-a_{T})^{\frac{1}{\phi_{A}}} Q_{NA,t}(x)^{\frac{\phi_{A}-1}{\phi_{A}}} \end{array}\right)^{\frac{\rho_{A}}{\rho_{A}-1}\frac{\phi_{A}-1}{\phi_{A}}} \right)^{\frac{\phi_{A}}{\rho_{A}-1}\frac{\phi_{A}}{\phi_{A}}}$$

where Q_{HA} , Q_{FA} and Q_{NA} are bundles of respectively Home tradable, Foreign tradable and Home non-tradable intermediate goods, $\rho > 0$ is the elasticity of substitution between tradables and $\phi > 0$ is the elasticity of substitution between tradable and non-tradable goods. The parameter a_H $(0 < a_H < 1)$ is the weight of domestic tradable, a_T $(0 < a_T < 1)$ the weight of tradable goods.

The production of investment good is similar. There are symmetric Home firms under perfect competition indexed by $y \in (0, s]$, and symmetric Foreign firms by $y^* \in (s, 1]$. Output of Home firm y is:

$$E_{t}(y) \equiv \left(\begin{array}{c} v_{T}^{\frac{1}{\phi_{E}}} \left(v_{H}^{\frac{1}{\rho_{E}}} Q_{HE,t}(y)^{\frac{\rho_{E}-1}{\rho_{E}}} + (1-v_{H})^{\frac{1}{\rho_{E}}} Q_{FE,t}(y)^{\frac{\rho_{E}-1}{\rho_{E}}} \right)^{\frac{\rho_{E}}{\rho_{E}-1}\frac{\phi_{E}-1}{\phi_{E}}} \\ + (1-v_{T})^{\frac{1}{\phi_{E}}} Q_{NE,t}(y)^{\frac{\phi_{E}-1}{\phi_{E}}} \end{array}\right)^{\frac{-\rho_{E}}{\rho_{E}}\frac{\phi_{E}-1}{\phi_{E}}} \right)^{\frac{-\rho_{E}}{\rho_{E}}\frac{\phi_{E}-1}{\phi_{E}}}$$

Finally, we assume that public expenditure C^g has the same composition as that of private consumption.

B Intermediate goods

B.1 Demand

Bundles used to produce the final consumption goods are CES indexes of differentiated intermediate goods, each produced by a single firm under conditions of monopolistic competition:

$$Q_{HA}(x) \equiv \left[\left(\frac{1}{s}\right)^{\theta_T} \int_0^s Q(h,x)^{\frac{\theta_T - 1}{\theta_T}} dh \right]^{\frac{\theta_T}{\theta_T - 1}} \tag{5}$$

$$Q_{FA}\left(x^{*}\right) \equiv \left[\left(\frac{1}{1-s}\right)^{\theta_{T}} \int_{s}^{1} Q\left(f,x\right)^{\frac{\theta_{T}-1}{\theta_{T}}} df\right]^{\frac{\theta_{T}}{\theta_{T}-1}} \tag{6}$$

$$Q_{NA}(x) \equiv \left[\left(\frac{1}{s}\right)^{\theta_N} \int_0^s Q(n,x)^{\frac{\theta_N - 1}{\theta_N}} dn \right]^{\frac{\theta_N}{\theta_T - 1}}$$
(7)

where firms in the Home tradable and non-tradable intermediate sectors and in the Foreign intermediate tradable sector are respectively indexed by $h \in (0, s), n \in (0, s), f \in (s, 1]$. Parameters $\theta_T, \theta_N > 1$ are respectively the elasticity of substitution between brands in the tradable and non-tradable sector. The prices of the non-tradable intermediate goods are denoted p(n). Each firm x takes these prices as given when minimizing production costs of the final good. The resulting demand for non-tradable intermediate input n is:

$$Q_{A,t}(n,x) = \left(\frac{1}{s}\right) \left(\frac{P_t(n)}{P_{N,t}}\right)^{-\theta_N} Q_{NA,t}(x)$$
(8)

where $P_{N,t}$ is the cost-minimizing price of one basket of local intermediates:

$$P_{N,t} = \left[\int_0^s P_t\left(n\right)^{1-\theta_N} dn\right]^{\frac{1}{1-\theta_N}} \tag{9}$$

We can derive $Q_A(h, x)$, $Q_A(f, x)$, $C_A^g(h, x)$, $C_A^g(f, x)$, P_H and P_F in a similar way. Firms y producing the final investment goods have similar demand curves. Aggregating over x and y, it can be shown that total demand for intermediate non-tradable good n is:

$$\int_{0}^{s} Q_{A,t}(n,x) \, dx + \int_{0}^{s} Q_{E,t}(n,y) \, dy + \int_{0}^{s} C_{t}^{g}(n,x) \, dx \qquad (10)$$

$$= \left(\frac{P_t(n)}{P_{N,t}}\right)^{-\theta_N} \left(Q_{NA,t} + Q_{NE,t} + C_{N,t}^g\right)$$
(11)

where C_N^g is non-tradable component of the public sector consumption. Home demands for Home and Foreign tradable intermediate goods can be derived in a similar way.

B.2 Supply

The supply of each Home non-tradable intermediate good n is denoted by $N^{S}(n)$:

$$N_{t}^{S}(n) = \left((1 - \alpha_{N})^{\frac{1}{\xi_{N}}} L_{N,t}(n)^{\frac{\xi_{N}-1}{\xi_{N}}} + \alpha^{\frac{1}{\xi_{N}}} K_{N,t}(n)^{\frac{\xi_{N}-1}{\xi_{N}}} \right)^{\frac{\xi_{N}}{\xi_{N}-1}}$$
(12)

Firm n uses labor $L_{N,t}^{p}(n)$ and capital $K_{N,t}(n)$ with constant elasticity of input substitution $\xi_{N} > 0$ and capital weight $0 < \alpha_{N} < 1$. Firms producing intermediate goods take the prices of labor inputs and capital as given. Denoting W_{t} the nominal wage index and R_{t}^{K} the nominal rental price of capital, cost minimization implies:

$$L_{N,t}^{p}(n) = (1 - \alpha_{N}) \left(\frac{W_{t}}{MC_{N,t}(n)}\right)^{-\xi_{N}} N_{t}^{S}(n)$$

$$K_{N,t}(n) = \alpha \left(\frac{R_{t}^{K}}{MC_{N,t}(n)}\right)^{-\xi_{N}} N_{t}^{S}(n)$$
(13)

where $MC_{N,t}(n)$ is the nominal marginal cost:

$$MC_{N,t}(n) = \left((1-\alpha) W_t^{1-\xi_N} + \alpha \left(R_t^K \right)^{1-\xi_N} \right)^{\frac{1}{1-\xi_N}}$$
(14)

The productions of each Home tradable good, $T^{S}(h)$, is similarly characterized.

B.3 Price setting in the intermediate sector

Consider now profit maximization in the Home country's nontradable intermediate sector. Each firm n sets the price $p_t(n)$ by maximizing the present discounted value of profits subject to demand constraint (10) and the quadratic adjustment costs:

$$AC_{N,t}^{p}(n) \equiv \frac{\kappa_{N}^{p}}{2} \left(\frac{P_{t}(n)}{P_{t-1}(n)} - 1\right)^{2} Q_{N,t} \quad \kappa_{N}^{p} \ge 0$$

paid in unit of sectorial product $Q_{N,t}$ and where κ_N^p measures the degree of price stickiness. The resulting first-order condition, expressed in terms of domestic consumption, is:

$$p_t(n) = \frac{\theta_N}{\theta_N - 1} mc_t(n) - \frac{A_t(n)}{\theta_N - 1}$$
(15)

where $mc_t(n)$ is the real marginal cost and A(n) contains terms related to the presence of price adjustment costs:

$$A_{t}(n) \approx \kappa_{N}^{p} \frac{P_{t}(n)}{P_{t-1}(n)} \left(\frac{P_{t}(n)}{P_{t-1}(n)} - 1\right)$$
(16)

$$-\beta \kappa_N^p \frac{P_{t+1}(n)}{P_t(n)} \left(\frac{P_{t+1}(n)}{P_t(n)} - 1\right) \frac{Q_{N,t+1}}{Q_{N,t}}$$
(17)

The above equations clarify the link between imperfect competition and nominal rigidities. As emphasized by Bayoumi et al.(2004), when the elasticity of substitution θ_N is very large and hence the competition in the sector is high, prices closely follow marginal costs, even though adjustment costs are large. To the contrary, it may be optimal to maintain stable prices and accommodate changes in demand through supply adjustments when the average markup over marginal costs is relatively high. If prices were flexible, optimal pricing would collapse to the standard pricing rule of constant markup over marginal costs (expressed in units of domestic consumption):

$$p_t(n) = \frac{\theta_N}{\theta_N - 1} m c_{N,t}(n) \tag{18}$$

Firms operating in the intermediate tradable sector solve a similar problem. We assume that there is market segmentation. Hence the firm producing the brand h chooses $p_t(h)$ in the Home market and $p_t^*(h)$ in the Foreign market as to maximize the expected flow of profits (in terms of domestic consumption units):

$$E_{t} \sum_{\tau=t}^{\infty} \Lambda_{t,\tau} \left[p_{\tau}(h) y_{\tau}(h) + p_{\tau}^{*}(h) y_{\tau}^{*}(h) - mc_{H,\tau}(h) \left(y_{\tau}(h) + y_{\tau}^{*}(h) \right) \right]$$

subject to quadratic price adjustment costs similar to those considered for nontradables and standard demand constraints. The term E_t denotes the expectation operator conditional on the information set at time t, $\Lambda_{t,\tau}$ is the appropriate discount rate and $m_{CH,t}(h)$ is the real marginal cost. The first order conditions with respect to $p_t(h)$ and $p_t^*(h)$ are:

$$p_t(h) = \frac{\theta_T}{\theta_T - 1} mc_t(h) - \frac{A_t(h)}{\theta_T - 1}$$
(19)

$$p_t^*(h) = \frac{\theta_T^*}{\theta_T - 1} mc_t(h) - \frac{A_t^*(h)}{\theta_T - 1}$$

$$\tag{20}$$

where θ_T^* is the elasticity of substitution of tradable intermediate goods in the Foreign country, while A(h) and $A^*(h)$ involve terms related to the presence of price adjustment costs:

$$A_t(h) \approx \kappa_H^p \frac{P_t(h)}{P_{t-1}(h)} \left(\frac{P_t(h)}{P_{t-1}(h)} - 1\right)$$
(21)

$$-\beta \kappa_{H}^{p} \frac{P_{t+1}(h)}{P_{t}(h)} \left(\frac{P_{t+1}(h)}{P_{t}(h)} - 1\right) \frac{Q_{H,t+1}}{Q_{H,t}}$$
(22)

$$A_{t}^{*}(h) \approx \theta_{T}^{*} - 1 + \kappa_{H}^{p} \frac{P_{t}^{*}(h)}{P_{t-1}^{*}(h)} \left(\frac{P_{t}^{*}(h)}{P_{t-1}^{*}(h)} - 1\right)$$
(23)

$$-\beta \kappa_{H}^{p} * \frac{P_{t+1}^{*}(h)}{P_{t}^{*}(h)} \left(\frac{P_{t+1}^{*}(h)}{P_{t}^{*}(h)} - 1\right) \frac{Q_{H,t+1}^{*}}{Q_{H,t}^{*}}$$
(24)

where $\kappa_H^p > 0$ ($\kappa_H^{p*} > 0$) measure the degree of nominal rigidity in the Home (Foreign) country. If nominal rigidities in the (domestic) export market are highly relevant (that is, if is relatively large), the degree of inertia of Home goods prices in the Foreign market will be high. If prices were flexible $(\kappa_H^p = \kappa_H^{p*})$ and $\theta_T = \theta_T^*$, then optimal price setting would be consistent with the cross-border law of one price:

$$p_t(h) = \frac{\theta_T}{\theta_T - 1} mc_t(h) = p_t^*(h)$$
(25)

B.4 Labor Market

In the case of firms in the nontradable intermediate sector, the labor input $L_N(n)$ is a CES combination of differentiated labor inputs supplied by domestic agents and defined over a continuum of mass equal to the country size $(j \in [0, s])$:

$$L_{N,t}\left(n\right) \equiv \left(\frac{1}{s}\right)^{\frac{1}{\psi}} \left[\int_{0}^{s} L_{t}\left(n,j\right)^{\frac{\psi-1}{\psi}} dj\right]^{\frac{\psi}{\psi-1}}$$
(26)

where L(n, j) is the demand of the labor input of type j by the producer of good n and $\psi > 1$ is the elasticity of substitution among labor inputs. Cost minimization implies:

$$L_t^p(n,j) = \left(\frac{1}{s}\right) \left(\frac{W_t(j)}{W_t}\right)^{-\psi} L_{N,t}^p(j), \qquad (27)$$

where W(j) is the nominal wage of labor input j and the wage index W is:

$$W_t = \left[\left(\frac{1}{s}\right) \int_0^s W_t \left(h\right)^{1-\psi} dj \right]^{\frac{1}{1-\psi}}.$$
 (28)

Similar equations hold for firms producing intermediate tradable goods. Each household is the monopolistic supplier of a labor input j and sets the nominal wage facing a downward-sloping demand, obtained by aggregating demand across Home firms. The wage adjustment is sluggish because of quadratic costs paid in terms of the total wage bill:

$$AC_t^W = \frac{\kappa_W}{2} \left(\frac{W_t}{W_{t-1}} - 1\right)^2 W_t L_t \tag{29}$$

where the parameter $\kappa_W > 0$ measures the degree of nominal wage rigidity and L is the total amount of labor in the Home economy.

B Monetary Policy

The monetary authority controls the short-term rate according to a Taylor rule of the form:

$$\left(\frac{1+i_t}{1+i}\right) = \left(\frac{1+i_{t-1}}{1+i}\right)^{\rho_i} \left(\Pi_{MU,t}\right)^{(1-\rho_i)\rho_\pi} \left(\frac{GDP_{MU,t}}{GDP_{MU,t-1}}\right)^{(1-\rho_i)\rho_{GDP}} \tag{30}$$

The parameter ρ_i ($0 < \rho_i < 1$) captures inertia in interest rate setting, while parameters ρ_{π} and ρ_{GDP} are respectively the weights of currency union's CPI inflation rate $\Pi_{MU,t}$ and GDP $GDP_{MU,t}$. The CPI inflation rate is a geometric average of CPI inflation rates in the Home and Foreign country (respectively Π_t and Π_t^*) with weights equal to the correspondent country size:

$$\Pi_{MU,t} \equiv (\Pi_t)^s \, (\Pi_t^*)^{1-s} \tag{31}$$

The union-wide GDP is the sum of the Home and Foreign GDPs (respectively GDP_t and GDP_t^*), both evaluated at the steady state prices:

$$GDP_{MU,t} \equiv GDP_t + rer * GDP_t^* \tag{32}$$

where *rer* is the Home real exchange rate, defined as the ratio of rest of the euro area to Home consumer prices.

C Market Clearing

The model is closed by imposing the following resource constraints and market clearing conditions. The resource constraint for Home nontradable final consumption good is:

$$\int_{0}^{s} A_{t}(x) dx \ge \int_{0}^{s} C_{t}(j) dj + C_{t}^{g}$$
(33)

The resource constraint for Home nontradable final investment good is:

$$\int_0^s E_t(x) \, dx \ge \int_0^s I_t(j) \, dj \tag{34}$$

The resource constraint for good n is

$$N_t^S(n) \ge \int_0^s Q_t(n, x) \, dx \tag{35}$$

The Home tradable h can be used by Home firms or imported by Foreign firms:

$$T_t^S(h) \ge \int_0^s Q_t(h, x) \, dx + \int_s^1 Q_t(h, x^*) \, dx^*$$
(36)

The resource constraints for factor market are:

$$\int_{0}^{s} L_{t}(j) \, dj \ge \int_{0}^{s} L_{t}(n) \, dn + \int_{0}^{s} L_{t}(h) \, dh + L_{t}^{g} \tag{37}$$

$$\int_{0}^{s} K_{t-1}(j) \, dj \ge \int_{0}^{s} K_{t}(n) \, dn + \int_{0}^{s} K_{t}(h) \, dh \tag{38}$$

The bond market clearing condition is:

$$\int_{0}^{s} B_{t}(j) \, dj + \int_{s}^{1} B_{t}(j^{*}) \, dj^{*} + B_{t}^{g} + B_{t}^{g,*} = 0 \tag{39}$$

D The equilibrium

We find a symmetric equilibrium of the model. In each country there is a representative agent and four representative sectorial firms (in the intermediate tradable sector, intermediate nontradable sector, consumption production sector and investment production sector). The equilibrium is a sequence of allocations and prices such that, given initial conditions and the sequence of exogenous shocks, each private agent and firm satisfy the correspondent first order conditions, the private and public sector budget constraints and market clearing conditions for goods, labor, capital and bond holdings.

		Rest of the
Parameter	Italy	Euro Area
Rate of time preference $(1/\beta^4 - 1) * 100$	5.0	5.0
Intertemporal elasticity of substitution $1/\sigma$	1.0	1.0
Frisch elasticity of labor $1/(\tau - 1)$	2.0	2.0
Depreciation rate of (private and public) capital δ, δ^*	0.025	0.025
Substitution between private and public goods in cons. bundle θ	1.5	1.5
Bias towards private goods in cons. bundle ω	0.8	0.8
Tradable Intermediate Goods		
Substitution between factors of production ξ_T, ξ_T^*	0.85	0.9
Bias towards capital α_T, α_T^*	0.75	0.7
Non tradable Intermediate Goods		
Substitution between factors of production ξ_N, ξ_N^*	0.79	0.95
Bias towards capital α_N	0.70	0.70
Production function of the public good		
Substitution between factors of production α_g	0.79	0.79
Bias towards intermediate goods $\gamma_{C^g}, \gamma_{C^g}^*$	0.15	0.15
Bias towards public employment $\gamma_{L^g}, \gamma_{L^g}^*$	0.15	0.15
Final consumption goods		
Substitution between domestic and imported goods ϕ_A, ϕ_A^*	1.5	1.5
Bias towards domestic tradable goods a_H, a_F^*	0.3	0.7
Substitution between domestic tradables and non tradables ρ_A, ρ_A^*	0.5	0.5
Bias towards tradable goods a_T, a_T^*	0.55	0.5
Final investment goods		
Substitution between domestic and imported goods ϕ_E, ϕ_E^*	1.5	1.5
Bias towards domestic tradable goods v_H, v_F^*	0.3	0.7
Substitution between domestic tradables and non tradables ρ_E, ρ_E^*	0.50	0.50
Bias towards tradable goods v_T, v_T^*	0.55	0.50
Size n and $(1-n)$	0.20	0.80

Table 1. Parametrization of Italy and the rest of the Euro Area(Base-Case Parameters)

 Table 2. Gross Markups

	Markups an	d Elasticities of S	Substitution
	Tradables	Non-tradables	Wages
Italy	1.2 $(\theta_T = 6)$	1.2 $(\theta_N = 6)$	$1.2 \ (\psi = 6)$
Rest of the euro area	$1.2 \ (\theta_T^*=6)$	$1.2 \ (\theta_N^*=6)$	$1.2 \ (\psi^*=6)$

 Table 3. Real and Nominal Adjustment Costs (Base-Case Parameters)

Parameter ("*" refers to rest of the Euro area)	Italy	Rest of the Euro Area
Real Adjustment Costs		
Investment ϕ_I, ϕ_I^*	1.00	1.00
Households' financial net position ϕ_{b1}, ϕ_{b2}	0.01, 0.01	-
Nominal Adjustment Costs		
Wages κ_W, κ_W^*	60	60
Price of domestically-produced tradables κ_H, k_F^*	60	60
Price of non tradables κ_N , κ_N^*	60	60
Price of imported intermediate goods κ_F , κ_H^*	60	60

 Table 4. Fiscal and Monetary Policy Rules

Parameter	IT	RoEA	EA
Fiscal policy rule			
ϕ_1,ϕ_1^*	± 1.5	± 1.5	-
$\phi_2,\phi_2^*,\phi_3,\phi_3^*$	± 15	± 15	-
Common monetary policy rule	-	-	
Lagged interest rate at t-1 ρ_i	-	-	0.9
Inflation ρ_{Π}	-	-	1.7
GDP growth ρ_{GDP}	-	-	0.4

e-Case I	Parameter	$\mathbf{s})$	
It	aly	Rest of	the Euro Area
data	model	data	model
59.7	56.8	57.1	59.5
20.7	14.2	21.1	19.8
25.8	23.6	-	-
25.9	23.6	-	-
0.0	0.0	-	-
9.3	9.3	10.3	10.3
17.1	16.7	16.1	18.3
11.0	11.9	10.1	10.1
4.6	5.3	2.5	3.0
44.3	45.6	41.6	44.3
	105.0		60.0
43.1	43.1	38.7	38.7
29.0	29.0	30.1	30.1
16.9	16.9	19.2	19.2
	-Case I It data 59.7 20.7 25.8 25.9 0.0 9.3 17.1 11.0 4.6 44.3 43.1 29.0 16.9	$\begin{array}{r c c c c c c c c c c c c c c c c c c c$	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

Table 5. Great Ratios and tax rates

Data sources: National Account data for macroeconomic variables (2008 values).

For fiscal variables: expenditure data (2008 values) are from AMECO database. Tax rates (in percent) are taken from Eurostat (2008). Macro and fiscal variables are expressed as a ratio to GDP.

Table 6. Steady state compart	isons: red	uction in t	ax and ex	penditure d	istortions (% changes)
	$^{3}\mathrm{L}$	ax distortion	JS	Expen	diture distortions
	<u> </u>	1% of GDP)	1% of GDP)
	$[Tr, au^\ell]$	$[Tr, \tau^k]$	$[Tr, au^c]$	$[Tr, C^g]$	$[Tr, L^g]$
	(1)	$(2)^{-1}$	(3)	(4)	(5)
Italy					
GDP	1.8	1.7	0.7	-0.6	-0.6
Private consumption (C)	2.3	1.1	0.9	0.8	0.7
Public good (Y_g)	0.2	0.2	0.1	-1.5	-3.4
Investment	2.5	4.6	0.9	-0.8	0.5
Export	0.3	0.3	0.1	-0.1	0.1
Import	3.1	2.8	1.1	-1.0	0.6
Hours worked (L)	2.4	0.5	0.9	-0.8	-1.3
Real wage (w)	-1.0	1.4	-0.4	0.3	-0.2
After-tax real wage $((1 - \tau^{\ell})w)$	3.4	1.4	-0.4	0.3	-0.2
Terms of trade	2.0	1.8	0.7	-0.6	0.4
Welfare	1.2	0.8	0.5	0.7	0.4
Rest of the Euro area					
GDP	0.6	0.6	0.2	-0.2	0.1
Private Consumption (C^*)	0.6	0.6	0.2	-0.2	0.1

÷ 1:1 ζ ¢ e

4	No change	in expenditures	No ché	ange in tax	rates	Reduction	n in labor	tax rate	Reduction	n in capita	l tax rate	Reduction	i in both t	ax rates
	B, τ^{ℓ}	$[B, au_k]$	$[B, C^g]$	$[B, L^g]$	[B,Tr]	$[B, C^g]$	$[B, L^g]$	[B,Tr]	$[B, C^g]$	$[B, L^g]$	[B,Tr]	$[B, C^g]$	$[B, L^g]$	[B,Tr]
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
Italy														
GDP	1.1	1.0	0.4	0.4	0.0	2.8	2.7	3.6	2.7	2.6	3.6	5.2	5.1	7.4
Private consumption (C)	1.4	0.7	-0.5	-0.4	0.0	5.8	5.5	4.6	3.7	3.5	2.4	10.2	9.6	7.1
Public good (Y_g)	0.1	0.1	0.8	1.9	0.0	-1.7	-4.5	0.5	-2.0	-5.0	0.5	-5.5	-12.9	0.9
Investment	1.5	2.7	0.5	-0.3	0.0	3.7	5.6	4.9	8.6	10.8	10.0	12.2	17.4	15.3
Export	0.2	0.2	0.1	0.0	0.0	0.4	0.6	0.6	0.4	0.7	0.6	0.8	1.3	1.1
Import	1.8	1.6	0.6	-0.3	0.0	4.6	6.9	6.0	4.4	7.0	6.0	8.6	14.9	12.4
Hours worked (L)	1.4	0.3	0.5	0.8	0.0	3.6	2.8	4.7	0.0	-1.0	1.2	3.1	1.1	5.9
Real wage (w)	-0.6	0.8	-0.2	0.1	0.0	-1.5	-2.2	-1.9	3.4	2.5	2.9	2.1	0.1	1.0
After-tax real wage $((1 - \tau^{\ell})w)$	2.0	0.8	-0.2	0.1	0.0	7.2	6.4	6.7	3.4	2.5	2.9	11.0	8.9	9.8
Terms of trade	1.2	1.1	0.4	-0.2	0.0	3.0	4.4	3.9	2.8	4.5	3.9	5.5	9.4	7.9
Welfare	0.7	0.5	-0.4	-0.3	0.0	3.4	2.9	2.4	2.8	2.3	1.7	6.5	5.3	4.1
Rest of the Euro area														
GDP	0.4	0.3	0.1	-0.1	0.0	1.0	1.4	1.2	0.9	1.4	1.2	1.8	3.0	2.5
Private Consumption (C^*)	0.4	0.4	0.1	-0.1	0.0	1.0	1.4	1.3	0.9	1.5	1.3	1.8	3.0	2.5

 Table 7. Steady state comparisons (% changes)

.

Table 8. Welfare along the	transitio	n (% changes)
No change in expenditures	$\left[B, \tau^{\ell}\right]$	-0.2
	$\left[B, \tau^{k}\right]$	0.0
No change in tax rates	$[B, C^g]$	-0.1
	$[B, L^g]$	-0.1
	[B,Tr]	0.0
Reduction in labor tax rate	$[B, C^g]$	2.8
	$[B, L^g]$	2.0
	[B,Tr]	1.5
Reduction in capital tax rate	$[B, C^g]$	2.0
	$[B, L^g]$	1.3
	[B,Tr]	0.7
Reduction in both tax rates	$[B, C^g]$	4.7
	$[B, L^g]$	3.1
	[B,Tr]	2.2

Lable 9. Kobustness, steady state	te compai	ISONS (% C	changes)												
		Baseline			$\tau = 3$			$\omega = 0.5$	_		$\theta = 0.8$			$\lambda = 0.35$	
	$[B, C^g]$	$[B, L^g]$	[B, Tr]	$[B, C^g]$	$[B, L^g]$	[B,Tr]	$[B, C^g]$	$[B, L^g]$	[B,Tr]	$[B, C^g]$	$[B, L^g]$	[B, Tr]	$[B, C^g]$	$[B, L^g]$	[B, Tr]
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)	(15)
Italy															
GDP	5.2	5.1	7.4	4.0	5.0	5.3	5.6	5.6	7.5	4.3	3.8	7.0	5.3	4.8	7.1
Private consumption (C)	10.2	9.6	7.1	8.7	9.1	4.2	10.8	10.2	7.4	9.4	8.3	6.7	10.4	9.4	6.8
Public good (Y_g)	-5.5	-12.9	0.9	-5.6	-23.3	0.6	-5.3	-10.1	1.0	-5.9	-12.7	0.9	-5.4	-11.3	0.9
Non Ricardian consumption (C^{NR})									_				11.6	8.2	4.9
Investment	12.2	17.4	15.3	10.4	17.0	12.1	12.9	18.1	15.7	11.0	15.9	14.9	12.4	17.2	15.0
Export	0.8	1.3	1.1	0.6	1.2	0.7	1.0	1.6	1.3	0.6	1.0	0.9	0.8	1.3	1.1
Import	8.6	14.9	12.4	6.6	14.4	8.5	9.4	15.6	12.7	7.2	13.1	11.9	8.9	14.5	12.0
Hours worked (L)	3.1	1.1	5.9	1.7	0.7	3.3	3.6	1.7	5.9	2.0	-0.5	5.5	3.2	0.8	5.5
Real wage (w)	2.1	0.1	1.0	2.7	0.2	2.1	1.9	-0.1	0.9	2.5	0.5	1.0	2.0	0.2	1.1
After-tax real wage $((1 - \tau^{\ell})w)$	11.0	8.9	9.8	11.7	9.1	11.1	10.8	8.7	9.8	11.5	9.4	9.9	11.0	0.0	9.9
Terms of trade	5.5	9.4	7.9	4.3	9.0	5.5	6.0	9.8	8.0	4.7	8.5	7.7	5.7	9.2	7.7
Welfare: steady state	6.5	5.3	4.1	6.0	3.8	2.7	4.2	2.5	3.6	5.0	3.1	3.7	6.6	0.0	4.8
Welfare: with transition	4.7	3.1	2.2	4.5	1.4	1.2	2.6	0.6	2.0	3.3	1.0	2.0	4.8	3.8	2.9
Rest of the Euro area									-						
GDP	1.8	3.0	2.5	1.3	2.8	1.7	2.0	3.3	2.7	1.4	2.5	2.2	1.8	2.9	2.4
Private Consumption (C^*)	1.8	3.0	2.5	1.3	2.8	1.7	2.0	3.4	2.8	1.4	2.5	2.3	1.8	3.0	2.5

Put

.



Figure 1. Welfare (% deviation from steady state), compensating the tax cuts with different expenditure items

capital income tax rate



Figure 2. Baseline scenario: fiscal variables



Figure 3. Baseline scenario: macroeconomic variables

- N. 726 Low skilled immigration and the expansion of private schools, by Davide Dottori and I-Ling Shen (October 2009).
- N. 727 Sorting, reputation and entry in a market for experts, by Enrico Sette (October 2009).
- N. 728 *Ricardian selection*, by Andrea Finicelli, Patrizio Pagano and Massimo Sbracia (October 2009).
- N. 729 *Trade-revealed TFP*, by Andrea Finicelli, Patrizio Pagano and Massimo Sbracia (October 2009).
- N. 730 The riskiness of corporate bonds, by Marco Taboga (October 2009).
- N. 731 *The interbank market after august 2007: what has changed and why?*, by Paolo Angelini, Andrea Nobili and Maria Cristina Picillo (October 2009).
- N. 732 *Tax morale and public spending inefficiency,* by Guglielmo Barone and Sauro Mocetti (November 2009).
- N. 733 Dynamic macroeconomic effects of public capital: evidence from regional Italian data, by Valter Di Giacinto, Giacinto Micucci and Pasqualino Montanaro (November 2009).
- N. 734 *Networks with decreasing returns to linking*, by Filippo Vergara Caffarelli (November 2009).
- N. 735 *Mutual guarantee institutions and small business finance*, by Francesco Columba, Leonardo Gambacorta and Paolo Emilio Mistrulli (November 2009).
- N. 736 Sacrifice ratio or welfare gain ratio? Disinflation in a DSGE monetary model, by Guido Ascari and Tiziano Ropele (January 2010).
- N. 737 The pro-competitive effect of imports from China: an analysis of firm-level price data, by Matteo Bugamelli, Silvia Fabiani and Enrico Sette (January 2010).
- N. 738 *External trade and monetary policy in a currency area*, by Martina Cecioni (January 2010).
- N. 739 The use of survey weights in regression analysis, by Ivan Faiella (January 2010).
- N. 740 Credit and banking in a DSGE model of the euro area, by Andrea Gerali, Stefano Neri, Luca Sessa and Federico Maria Signoretti (January 2010).
- N. 741 Why do (or did?) banks securitize their loans? Evidence from Italy, by Massimiliano Affinito and Edoardo Tagliaferri (January 2010).
- N. 742 *Outsourcing versus integration at home or abroad*, by Stefano Federico (February 2010).
- N. 743 *The effect of the Uruguay round on the intensive and extensive margins of trade*, by Ines Buono and Guy Lalanne (February 2010).
- N. 744 Trade, technical progress and the environment: the role of a unilateral green tax on consumption, by Daniela Marconi (February 2010).
- N. 745 Too many lawyers? Litigation in Italian civil courts, by Amanda Carmignani and Silvia Giacomelli (February 2010).
- N. 746 On vector autoregressive modeling in space and time, by Valter Di Giacinto (February 2010).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- S. SIVIERO and D. TERLIZZESE, *Macroeconomic forecasting: Debunking a few old wives' tales,* Journal of Business Cycle Measurement and Analysis, v. 3, 3, pp. 287-316, **TD No. 395 (February 2001).**
- S. MAGRI, Italian households' debt: The participation to the debt market and the size of the loan, Empirical Economics, v. 33, 3, pp. 401-426, **TD No. 454 (October 2002).**
- L. CASOLARO. and G. GOBBI, *Information technology and productivity changes in the banking industry*, Economic Notes, Vol. 36, 1, pp. 43-76, **TD No. 489 (March 2004).**
- G. FERRERO, *Monetary policy, learning and the speed of convergence,* Journal of Economic Dynamics and Control, v. 31, 9, pp. 3006-3041, **TD No. 499 (June 2004).**
- M. PAIELLA, Does wealth affect consumption? Evidence for Italy, Journal of Macroeconomics, Vol. 29, 1, pp. 189-205, TD No. 510 (July 2004).
- F. LIPPI. and S. NERI, *Information variables for monetary policy in a small structural model of the euro area*, Journal of Monetary Economics, Vol. 54, 4, pp. 1256-1270, **TD No. 511 (July 2004).**
- A. ANZUINI and A. LEVY, *Monetary policy shocks in the new EU members: A VAR approach*, Applied Economics, Vol. 39, 9, pp. 1147-1161, **TD No. 514 (July 2004).**
- D. JR. MARCHETTI and F. Nucci, *Pricing behavior and the response of hours to productivity shocks*, Journal of Money Credit and Banking, v. 39, 7, pp. 1587-1611, **TD No. 524 (December 2004).**
- R. BRONZINI, *FDI Inflows, agglomeration and host country firms' size: Evidence from Italy*, Regional Studies, Vol. 41, 7, pp. 963-978, **TD No. 526 (December 2004).**
- L. MONTEFORTE, Aggregation bias in macro models: Does it matter for the euro area?, Economic Modelling, 24, pp. 236-261, **TD No. 534 (December 2004).**
- A. NOBILI, Assessing the predictive power of financial spreads in the euro area: does parameters instability matter?, Empirical Economics, Vol. 31, 1, pp. 177-195, **TD No. 544 (February 2005).**
- A. DALMAZZO and G. DE BLASIO, *Production and consumption externalities of human capital: An empirical study for Italy*, Journal of Population Economics, Vol. 20, 2, pp. 359-382, **TD No. 554 (June 2005).**
- M. BUGAMELLI and R. TEDESCHI, Le strategie di prezzo delle imprese esportatrici italiane, Politica Economica, v. 23, 3, pp. 321-350, TD No. 563 (November 2005).
- L. GAMBACORTA and S. IANNOTTI, Are there asymmetries in the response of bank interest rates to monetary shocks?, Applied Economics, v. 39, 19, pp. 2503-2517, TD No. 566 (November 2005).
- P. ANGELINI and F. LIPPI, *Did prices really soar after the euro cash changeover? Evidence from ATM withdrawals*, International Journal of Central Banking, Vol. 3, 4, pp. 1-22, **TD No. 581 (March 2006).**
- A. LOCARNO, Imperfect knowledge, adaptive learning and the bias against activist monetary policies, International Journal of Central Banking, v. 3, 3, pp. 47-85, **TD No. 590 (May 2006).**
- F. LOTTI and J. MARCUCCI, *Revisiting the empirical evidence on firms' money demand*, Journal of Economics and Business, Vol. 59, 1, pp. 51-73, **TD No. 595** (May 2006).
- P. CIPOLLONE and A. ROSOLIA, *Social interactions in high school: Lessons from an earthquake*, American Economic Review, Vol. 97, 3, pp. 948-965, **TD No. 596 (September 2006).**
- L. DEDOLA and S. NERI, *What does a technology shock do? A VAR analysis with model-based sign restrictions*, Journal of Monetary Economics, Vol. 54, 2, pp. 512-549, **TD No. 607 (December 2006).**
- F. VERGARA CAFFARELLI, *Merge and compete: strategic incentives for vertical integration*, Rivista di politica economica, v. 97, 9-10, serie 3, pp. 203-243, **TD No. 608 (December 2006).**
- A. BRANDOLINI, Measurement of income distribution in supranational entities: The case of the European Union, in S. P. Jenkins e J. Micklewright (eds.), Inequality and Poverty Re-examined, Oxford, Oxford University Press, TD No. 623 (April 2007).
- M. PAIELLA, The foregone gains of incomplete portfolios, Review of Financial Studies, Vol. 20, 5, pp. 1623-1646, TD No. 625 (April 2007).
- K. BEHRENS, A. R. LAMORGESE, G.I.P. OTTAVIANO and T. TABUCHI, Changes in transport and non transport costs: local vs. global impacts in a spatial network, Regional Science and Urban Economics, Vol. 37, 6, pp. 625-648, TD No. 628 (April 2007).

- M. BUGAMELLI, Prezzi delle esportazioni, qualità dei prodotti e caratteristiche di impresa: analisi su un campione di imprese italiane, v. 34, 3, pp. 71-103, Economia e Politica Industriale, TD No. 634 (June 2007).
- G. ASCARI and T. ROPELE, *Optimal monetary policy under low trend inflation*, Journal of Monetary Economics, v. 54, 8, pp. 2568-2583, **TD No. 647** (November 2007).
- R. GIORDANO, S. MOMIGLIANO, S. NERI and R. PEROTTI, *The Effects of Fiscal Policy in Italy: Evidence from a VAR Model*, European Journal of Political Economy, Vol. 23, 3, pp. 707-733, **TD No. 656** (January 2008).
- B. ROFFIA and A. ZAGHINI, *Excess money growth and inflation dynamics*, International Finance, v. 10, 3, pp. 241-280, **TD No. 657 (January 2008).**
- G. BARBIERI, P. CIPOLLONE and P. SESTITO, Labour market for teachers: demographic characteristics and allocative mechanisms, Giornale degli economisti e annali di economia, v. 66, 3, pp. 335-373, TD No. 672 (June 2008).
- E. BREDA, R. CAPPARIELLO and R. ZIZZA, Vertical specialisation in Europe: evidence from the import content of exports, Rivista di politica economica, v. 97, 3, pp. 189, **TD No. 682 (August 2008).**

- P. ANGELINI, *Liquidity and announcement effects in the euro area*, Giornale degli Economisti e Annali di Economia, v. 67, 1, pp. 1-20, **TD No. 451 (October 2002).**
- P. ANGELINI, P. DEL GIOVANE, S. SIVIERO and D. TERLIZZESE, Monetary policy in a monetary union: What role for regional information?, International Journal of Central Banking, v. 4, 3, pp. 1-28, TD No. 457 (December 2002).
- F. SCHIVARDI and R. TORRINI, *Identifying the effects of firing restrictions through size-contingent Differences in regulation*, Labour Economics, v. 15, 3, pp. 482-511, **TD No. 504 (June 2004).**
- L. GUISO and M. PAIELLA,, *Risk aversion, wealth and background risk*, Journal of the European Economic Association, v. 6, 6, pp. 1109-1150, **TD No. 483 (September 2003).**
- C. BIANCOTTI, G. D'ALESSIO and A. NERI, *Measurement errors in the Bank of Italy's survey of household income and wealth*, Review of Income and Wealth, v. 54, 3, pp. 466-493, **TD No. 520 (October 2004).**
- S. MOMIGLIANO, J. HENRY and P. HERNÁNDEZ DE COS, The impact of government budget on prices: Evidence from macroeconometric models, Journal of Policy Modelling, v. 30, 1, pp. 123-143 TD No. 523 (October 2004).
- L. GAMBACORTA, *How do banks set interest rates?*, European Economic Review, v. 52, 5, pp. 792-819, **TD No. 542 (February 2005).**
- P. ANGELINI and A. GENERALE, *On the evolution of firm size distributions*, American Economic Review, v. 98, 1, pp. 426-438, **TD No. 549 (June 2005).**
- R. FELICI and M. PAGNINI, *Distance, bank heterogeneity and entry in local banking markets*, The Journal of Industrial Economics, v. 56, 3, pp. 500-534, **No. 557 (June 2005).**
- S. DI ADDARIO and E. PATACCHINI, *Wages and the city. Evidence from Italy*, Labour Economics, v.15, 5, pp. 1040-1061, **TD No. 570 (January 2006).**
- M. PERICOLI and M. TABOGA, Canonical term-structure models with observable factors and the dynamics of bond risk premia, Journal of Money, Credit and Banking, v. 40, 7, pp. 1471-88, TD No. 580 (February 2006).
- E. VIVIANO, *Entry regulations and labour market outcomes. Evidence from the Italian retail trade sector*, Labour Economics, v. 15, 6, pp. 1200-1222, **TD No. 594 (May 2006).**
- S. FEDERICO and G. A. MINERVA, Outward FDI and local employment growth in Italy, Review of World Economics, Journal of Money, Credit and Banking, v. 144, 2, pp. 295-324, TD No. 613 (February 2007).
- F. BUSETTI and A. HARVEY, *Testing for trend*, Econometric Theory, v. 24, 1, pp. 72-87, **TD No. 614** (February 2007).

- V. CESTARI, P. DEL GIOVANE and C. ROSSI-ARNAUD, Memory for prices and the Euro cash changeover: an analysis for cinema prices in Italy, In P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 619 (February 2007).
- B. H. HALL, F. LOTTI and J. MAIRESSE, Employment, innovation and productivity: evidence from Italian manufacturing microdata, Industrial and Corporate Change, v. 17, 4, pp. 813-839, TD No. 622 (April 2007).
- J. SOUSA and A. ZAGHINI, *Monetary policy shocks in the Euro Area and global liquidity spillovers,* International Journal of Finance and Economics, v.13, 3, pp. 205-218, **TD No. 629 (June 2007).**
- M. DEL GATTO, GIANMARCO I. P. OTTAVIANO and M. PAGNINI, Openness to trade and industry cost dispersion: Evidence from a panel of Italian firms, Journal of Regional Science, v. 48, 1, pp. 97-129, TD No. 635 (June 2007).
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "inflation perceptions"? A survey-based analysis of Italian consumers, in P. Del Giovane e R. Sabbatini (eds.), The Euro Inflation and Consumers' Perceptions. Lessons from Italy, Berlin-Heidelberg, Springer, TD No. 655 (January 2008).
- R. BRONZINI, G. DE BLASIO, G. PELLEGRINI and A. SCOGNAMIGLIO, *La valutazione del credito d'imposta per gli investimenti*, Rivista di politica economica, v. 98, 4, pp. 79-112, **TD No. 661 (April 2008).**
- B. BORTOLOTTI, and P. PINOTTI, *Delayed privatization*, Public Choice, v. 136, 3-4, pp. 331-351, **TD No.** 663 (April 2008).
- R. BONCI and F. COLUMBA, Monetary policy effects: New evidence from the Italian flow of funds, Applied Economics, v. 40, 21, pp. 2803-2818, TD No. 678 (June 2008).
- M. CUCCULELLI, and G. MICUCCI, *Family Succession and firm performance: evidence from Italian family firms*, Journal of Corporate Finance, v. 14, 1, pp. 17-31, **TD No. 680 (June 2008).**
- A. SILVESTRINI and D. VEREDAS, *Temporal aggregation of univariate and multivariate time series models: a survey*, Journal of Economic Surveys, v. 22, 3, pp. 458-497, **TD No. 685 (August 2008).**

- F. PANETTA, F. SCHIVARDI and M. SHUM, Do mergers improve information? Evidence from the loan market, Journal of Money, Credit, and Banking, v. 41, 4, pp. 673-709, TD No. 521 (October 2004).
- M. BUGAMELLI and F. PATERNÒ, *Do workers' remittances reduce the probability of current account reversals?*, World Development, v. 37, 12, pp. 1821-1838, **TD No. 573 (January 2006).**
- P. PAGANO and M. PISANI, *Risk-adjusted forecasts of oil prices*, The B.E. Journal of Macroeconomics, v. 9, 1, Article 24, **TD No. 585 (March 2006).**
- M. PERICOLI and M. SBRACIA, The CAPM and the risk appetite index: theoretical differences, empirical similarities, and implementation problems, International Finance, v. 12, 2, pp. 123-150, TD No. 586 (March 2006).
- U. ALBERTAZZI and L. GAMBACORTA, *Bank profitability and the business cycle*, Journal of Financial Stability, v. 5, 4, pp. 393-409, **TD No. 601 (September 2006).**
- S. MAGRI, *The financing of small innovative firms: the Italian case*, Economics of Innovation and New Technology, v. 18, 2, pp. 181-204, **TD No. 640 (September 2007).**
- S. MAGRI, *The financing of small entrepreneurs in Italy*, Annals of Finance, v. 5, 3-4, pp. 397-419, **TD** No. 640 (September 2007).
- V. DI GIACINTO and G. MICUCCI, The producer service sector in Italy: long-term growth and its local determinants, Spatial Economic Analysis, Vol. 4, No. 4, pp. 391-425, TD No. 643 (September 2007).
- F. LORENZO, L. MONTEFORTE and L. SESSA, *The general equilibrium effects of fiscal policy: estimates for the euro area*, Journal of Public Economics, v. 93, 3-4, pp. 559-585, **TD No. 652 (November 2007).**
- R. GOLINELLI and S. MOMIGLIANO, *The Cyclical Reaction of Fiscal Policies in the Euro Area. A Critical Survey of Empirical Research*, Fiscal Studies, v. 30, 1, pp. 39-72, **TD No. 654 (January 2008).**
- P. DEL GIOVANE, S. FABIANI and R. SABBATINI, What's behind "Inflation Perceptions"? A survey-based analysis of Italian consumers, Giornale degli Economisti e Annali di Economia, v. 68, 1, pp. 25-52, TD No. 655 (January 2008).

- F. MACCHERONI, M. MARINACCI, A. RUSTICHINI and M. TABOGA, *Portfolio selection with monotone mean*variance preferences, Mathematical Finance, v. 19, 3, pp. 487-521, **TD No. 664 (April 2008).**
- M. AFFINITO and M. PIAZZA, What are borders made of? An analysis of barriers to European banking integration, in P. Alessandrini, M. Fratianni and A. Zazzaro (eds.): The Changing Geography of Banking and Finance, Dordrecht Heidelberg London New York, Springer, TD No. 666 (April 2008).
- G. FERRERO and A. NOBILI, *Futures contract rates as monetary policy forecasts*, International Journal of Central Banking, v. 5, 2, pp. 109-145, **TD No. 681 (June 2008).**
- L. ARCIERO, C. BIANCOTTI, L. D'AURIZIO and C. IMPENNA, *Exploring agent-based methods for the analysis* of payment systems: A crisis model for StarLogo TNG, Journal of Artificial Societies and Social Simulation, v. 12, 1, **TD No. 686 (August 2008).**
- A. CALZA and A. ZAGHINI, Nonlinearities in the dynamics of the euro area demand for M1, Macroeconomic Dynamics, v. 13, 1, pp. 1-19, **TD No. 690 (September 2008).**
- L. FRANCESCO and A. SECCHI, *Technological change and the households' demand for currency*, Journal of Monetary Economics, v. 56, 2, pp. 222-230, **TD No. 697 (December 2008).**
- G. ASCARI and T. ROPELE, *Trend inflation, taylor principle, and indeterminacy*, Journal of Money, Credit and Banking, v. 41, 8, pp. 1557-1584, **TD No. 708** (May 2007).
- S. COLAROSSI and A. ZAGHINI, Gradualism, transparency and the improved operational framework: a look at overnight volatility transmission, International Finance, v. 12, 2, pp. 151-170, TD No. 710 (May 2009).
- M. BUGAMELLI, F. SCHIVARDI and R. ZIZZA, *The euro and firm restructuring*, in A. Alesina e F. Giavazzi (eds): Europe and the Euro, Chicago, University of Chicago Press, **TD No. 716 (June 2009).**
- B. HALL, F. LOTTI and J. MAIRESSE, *Innovation and productivity in SMEs: empirical evidence for Italy*, Small Business Economics, v. 33, 1, pp. 13-33, **TD No. 718 (June 2009).**

S. MAGRI, *Debt maturity choice of nonpublic Italian firms*, Journal of Money, Credit, and Banking, v.42, 2-3, pp. 443-463, **TD No. 574 (January 2006).**

FORTHCOMING

- L. MONTEFORTE and S. SIVIERO, *The Economic Consequences of Euro Area Modelling Shortcuts*, Applied Economics, **TD No. 458 (December 2002).**
- M. BUGAMELLI and A. ROSOLIA, *Produttività e concorrenza estera*, Rivista di politica economica, **TD No.** 578 (February 2006).
- G. DE BLASIO and G. NUZZO, *Historical traditions of civicness and local economic development*, Journal of Regional Science, **TD No. 591 (May 2006).**
- R. BRONZINI and P. PISELLI, *Determinants of long-run regional productivity with geographical spillovers: the role of R&D, human capital and public infrastructure,* Regional Science and Urban Economics, **TD No. 597 (September 2006).**
- E. IOSSA and G. PALUMBO, *Over-optimism and lender liability in the consumer credit market*, Oxford Economic Papers, **TD No. 598 (September 2006).**
- S. NERI and A. NOBILI, *The transmission of US monetary policy to the euro area*, International Finance, **TD No. 606 (December 2006).**
- G. FERRERO, A. NOBILI and P. PASSIGLIA, Assessing Excess Liquidity in the Euro Area: The Role of Sectoral Distribution of Money, Applied Economics, **TD No. 627** (April 2007).
- A. CIARLONE, P. PISELLI and G. TREBESCHI, *Emerging Markets' Spreads and Global Financial Conditions*, Journal of International Financial Markets, Institutions & Money, **TD No. 637 (June 2007).**
- Y. ALTUNBAS, L. GAMBACORTA and D. MARQUÉS, *Securitisation and the bank lending channel*, European Economic Review, **TD No. 653 (November 2007).**
- M. IACOVIELLO and S. NERI, *Housing market spillovers: evidence from an estimated DSGE model,* American Economic Journal: Macroeconomics, **TD No. 659 (January 2008).**

- F. BALASSONE, F. MAURA and S. ZOTTERI, *Cyclical asymmetry in fiscal variables in the EU*, Empirica, **TD** No. 671 (June 2008).
- M. BUGAMELLI and F. PATERNÒ, *Output growth volatility and remittances*, Economica, **TD No. 673 (June 2008).**
- F. D'AMURI, O. GIANMARCO I.P. and P. GIOVANNI, *The labor market impact of immigration on the western* german labor market in the 1990s, European Economic Review, **TD No. 687 (August 2008).**
- A. ACCETTURO, Agglomeration and growth: the effects of commuting costs, Papers in Regional Science, **TD No. 688 (September 2008).**
- L. FORNI, A. GERALI and M. PISANI, *Macroeconomic effects of greater competition in the service sector: the case of Italy*, Macroeconomic Dynamics, **TD No. 706 (March 2009).**
- Y. ALTUNBAS, L. GAMBACORTA, and D. MARQUÉS-IBÁÑEZ, *Bank risk and monetary policy*, Journal of Financial Stability, **TD No. 712 (May 2009).**
- V. DI GIACINTO, G. MICUCCI and P. MONTANARO, Dynamic macroeconomic effects of public capital: evidence from regional Italian data, Giornale degli economisti e annali di economia, TD No. 733 (November 2009).
- F. COLUMBA, L. GAMBACORTA and P. E. MISTRULLI, *Mutual Guarantee institutions and small business finance,* Journal of Financial Stability, **TD No. 735 (November 2009).**
- L. FORNI, A. GERALI and M. PISANI, *The macroeconomics of fiscal consolidations in euro area countries,* Journal of Economic Dynamics and Control, **TD No. 747 (March 2010).**